Screening without Single Crossing

Changhyun Kwak^{*} Yale University

November 18, 2022

Job Market Paper (Click here for the latest version.)

Abstract

This paper provides a sufficient condition under which a general screening problem can be reduced, without the principal's payoff loss, to one with the single crossing property. The sufficient condition requires the agent's types to be ordered in a way that *two* marginal rates of substitution are both increasing in the type order. The monotonicity of one marginal rate of substitution allows the optimal mechanism to use only an allocation subset. The monotonicity of the other marginal rate of substitution ensures that the agent's preferences over the allocation subset satisfy the single crossing property. I apply the result to various economic problems with multi-dimensional allocations: multi-product monopoly, product line design, Bayesian persuasion, and delegation.

^{*}Department of Economics, Yale University, New Haven, CT 06520-8268; changhyun.kwak@yale.edu. I am indebted to my advisors, Larry Samuelson, Philipp Strack, and Kai Hao Yang, for their constant support and encouragement. I appreciate the helpful comments from Dirk Bergemann, Marina Halac, Johannes Hörner, Ravi Jagadeesan, Jihong Lee, Jaewon Lee, Niko Matouschek, Weicheng Min, and Suk Joon Son. I also thank seminar participants in Yale's Micro Theory Lunch seminar.

1 Introduction

1.1 Motivation

A key to overcoming the ubiquitous problem of asymmetric information rests in the well-known "single crossing property" (Milgrom and Shannon, 1994). It says that in the principal-agent relationship with hidden information, the agent types and allocations can be ordered so that higher types have increasingly stronger preferences for higher allocations. Thus, if the screening problem satisfies the single crossing property, every implementable choice rule has to satisfy "type-monotonicity"; that is, the optimal choice from every menu of allocations is monotone in type. Also, a choice rule can be implemented if only local incentive compatibility constraints according to the type order are satisfied, which provides a simple representation of the total rent as a function of the choice rule. Since the type order is independent of the choice rule, the representation simplifies the screening problem into a maximization problem of only choice rule subject to the monotonicity constraint.

The single crossing property, however, is often a strong assumption. In particular, it requires the agent's preferences to be ordered by a single-dimensional characteristic, which is inapplicable to multi-product monopoly and numerous other screening problems. In absence of single crossing, the set of binding constraints depends on the choice rule, and this makes the optimal mechanism difficult to predict. The optimal mechanism may take many different forms depending on the agent's type distribution. For example, in the single-product monopoly, it is well-known that single price is optimal (Riley and Zeckhauser, 1983). However if the monopolist sells more than two goods, the optimal selling mechanism ranges from the simplest pure bundling (Pavlov, 2011; Haghpanah and Hartline, 2021) and mixed bundling (Manelli and Vincent, 2006) to probabilistic bundling and even a continuum of lotteries (Thanassoulis, 2004; Hart and Nisan, 2019), depending on the distribution of consumers.

This paper seeks to go beyond single crossing. I ask the following question. When, in general, can a screening problem be reduced to one that satisfies the single crossing property? I call a screening problem reducible to a single crossing problem, or *single-crossing-reducible*, under a subset of allocations X if (i) the principal can restrict attention to X without payoff loss when screening the agent and (ii) the agent's preferences over X satisfy the single crossing property. In such situations, the optimal mechanism can be found by solving a much more tractable problem.

I derive a sufficient condition for single-crossing-reducibility characterized by the agent's type set. The sufficient condition requires the agent types to be totally ordered in a way that

two marginal rates of substitution are both increasing in the same type order. The monotonicity of one marginal rate of substitution ensures that the optimal mechanism can restrict attention to an allocation subset X. The monotonicity of the other marginal rate of substitution guarantees that the agent's preferences over X satisfy the single crossing property.

I apply this result to various economic settings to derive new sufficient conditions for the optimality of simple mechanisms as well as to generalize some previous sufficient conditions. Because all the sufficient conditions are only about the type set, the simple mechanisms are robustly optimal with respect to the distribution over the type set. First, in multiple good monopoly, I provide a sufficient condition for upgrade pricing, i.e., a menu of product bundles ordered by a set inclusion, which generalizes the optimality result of pure bundling in Haghpanah and Hartline (2021). The sufficient condition consists of monotone valuation and monotone MRS: Valuation for every good and MRS between each pair of the goods are monotone according to the same type order. Moreover, I show that if the valuation for each good is strictly monotone, monotone MRS is necessary and sufficient for the robust optimality of upgrade pricing with respect to the distribution over the type set.

In Bayesian persuasion, I extend the sufficient condition for the nested interval structure of acceptance sets in Guo and Shmaya (2019) to a larger class of payoff functions. In delegation, I characterize when no screening, i.e. offering the same cap regardless of the agent bias, is optimal. The corresponding sufficient condition is about the state distribution, and my result can also extend the Tanner (2018) condition about risk aversion to a more general form involving payoff functions.

The rest of the paper is organized as the follows. Section 2 introduces the model, followed by an illustrative example in Section 3. Section 4 provides a sufficient condition for the reducibility of screening problem to a single crossing problem. Section 5 uses the result to obtain sufficient conditions for the optimality of simple mechanisms that have nested forms. Section 6 discusses various applications in a unified setting and Section 7 concludes.

1.2 Literature Review

In the screening literature, the restrictiveness of the single crossing property is well-documented (Hörner, 2008) and its violation becomes inevitable especially when the agent types or allocations are multi-dimensional (Rochet and Stole, 2003). As explained in Rochet and Choné (1998), the violation makes it extremely difficult to generalize the analysis of the one-dimensional analogue. This has led the literature to focus its attention on the certification of the optimality of particular candidate mechanisms, and significant contributions have been made by using the

duality approach along this direction (Rochet and Choné, 1998; Cai, Devanur, and Weinberg, 2016; Daskalakis, Deckelbaum, and Tzamos, 2017; Carroll, 2017; Haghpanah and Hartline, 2021).

A branch of the screening literature has attempted to relax the single crossing property while minimizing the loss in tractability.¹ In the standard setting of one dimensional type/allocation with money transfers, Araujo and Moreira (2010) and Schottmüller (2015) allow a one-time violation of single crossing property and characterize when the solution has U-shaped and monotone solutions, respectively. In multi-dimensional screening, Carroll (2012) studies when local incentive compatibility (IC) constraints imply global ones, which is a desirable property of the single crossing property. In the current paper, I offer a condition under which downward IC constraints combined with optimality are sufficient for full incentive constraints.

In dealing with screening problems without single crossing property, the role of downward IC constraints dates back to Moore (1984) and Matthews and Moore (1987). With two dimensional allocations with money transfers, Matthews and Moore (1987) characterizes optimal mechanisms that can be non-monotonic, which implies that non-local IC constraints might be essential, unlike problems with single crossing property. In their paper, the key step is finding a sufficient condition about the preferences where a solution of the relaxed problem with only downward IC constraints is that of the original problem, which requires utility functions of type for every pair of allocations to cross at most twice in a particular fashion. Finding a sufficient condition for the equivalence of the two solutions is also a critical step in our analysis, but our condition takes a different form for the reduction to a screening problem with single crossing property.

There is also a body of papers decomposing a multi-dimensional type set into one dimensional paths with single crossing property so that solving the screening problem for each path is sufficient for obtaining the solution for the original problem (Armstrong, 1996; Eső and Szentes, 2007; Haghpanah and Hartline, 2021; Bergemann, Bonatti, Haupt, and Smolin, 2021; Yang, 2021). Among the aforementioned papers, Haghpanah and Hartline (2021) and Yang (2021) are closest to this paper. Haghpanah and Hartline (2021) finds a sufficient condition for the reduction of multiple good monopoly to single good monopoly with grand bundle. The result in this paper allows us to not only recover their condition but also derive the optimality condition of more general selling mechanisms such as upgrade pricing (see Section 6.1). Yang (2021) studies the effectiveness of the use of a socially wasteful component in a specific setting in which money transfers are allowed and preferences are additively separable. The paper uses

¹For a similar approach in signaling, see, for example, Chen, Ishida, and Suen (2022).

similar proof methods, but I study a more general screening problem to characterize when the problem can be reduced to a single crossing problem. Being specialized to the setting, the single-crossing-reducibility condition in this paper coincides with the optimality condition of Yang (2021) (see Remark 4).

2 Model

2.1 Setup

I first define a screening problem with a single principal and a single agent. A screening environment (Θ, Z, U) is defined by the agent's finite type space Θ , an allocation set Z, assumed to be a compact subset of a Euclidean space endowed with Lebesgue measure, and the principal and agent's utility functions $U_P(z, \theta), U_A(z, \theta) : Z \times \Theta \to \mathbb{R}$ that are continuous in $z \in Z$. Here, I do not allow any mixing between allocations, that is, only degenerate distributions over Z are allowed.² Also, note that we do not assume the quasilinearity or the independence of state about the players' preferences. Hence, it allows for every possible state dependence and complementarity between the components of elements of Z in the players' preferences. Lastly, for each type the agent's outside option value is normalized to be zero.

In the screening environment, the agent privately knows his type $\theta \in \Theta$ while the principal only has a prior belief $\mu \in \Delta \Theta$. The principal can screen the agent by using a mechanism.

2.2 Optimal IC mechanism

According to the revelation principle, I can restrict attention to direct mechanisms, functions of the agent's type, defined as $h : \Theta \to Z$. Throughout this text, a direct incentive compatible (IC) mechanism is a direct mechanism that satisfies both incentive compatibility (IC) and individual rationality (IR) constraints:

$$\begin{split} IC: \ U_A(h(\theta),\theta) \geq U_A(h(\theta'),\theta) & \qquad \text{for all } \theta,\theta'\in\Theta\\ IR: \ U_A(h(\theta),\theta) \geq 0 & \qquad \text{for all } \theta\in\Theta \end{split}$$

I denote the set of all the direct incentive compatible mechanisms as M. Then, the principal's problem is finding the *optimal IC mechanism under* Z which maximizes her own expected

²This assumption is with little loss of generality. For example, Z might be interpreted as the set of mixtures over a finite set of pure allocations.

utility among all the elements of M.

$$\max_{h \in M} \mathbb{E}[U_P(h(\theta), \theta)] \tag{1}$$

Next, I consider an alternative problem of the principal where the allocation set is restricted to be a compact subset X of Z. The setting is same except that the principal now has to choose every allocation from X for all type reports. Let M_X denote the set of direct incentive compatible mechanisms whose range is a subset of X. Then, the principal's problem can be written as the following by replacing M with M_X in (1):

$$\max_{h \in M_X} \mathbb{E}[U_P(h(\theta), \theta)]$$
(2)

I call the solution an *optimal IC mechanism under* X. The existence of both optimal IC mechanisms can be easily shown by the compactness argument.

2.3 Single crossing screening problem

Here, I define a class of screening problems with a desirable ordering property that makes them much more analytically tractable.

Definition 1 A screening problem (Θ, X, U) is a **single crossing** problem if there exists a total order \succeq_X on X and a total order \succeq_{Θ} on Θ s.t. for every $x' \succeq_X x$ and $\theta' \succeq_{\Theta} \theta$,

$$\begin{split} U_A(x',\theta) &\geq U_A(x,\theta) \Rightarrow U_A(x',\theta') \geq U_A(x,\theta') \\ U_A(x',\theta) &> U_A(x,\theta) \Rightarrow U_A(x',\theta') > U_A(x,\theta') \\ U_A(x',\theta) &\geq 0 \Rightarrow U_A(x',\theta') \geq 0 \end{split} \tag{P}$$

The first two inequalities imply that higher type has a stronger preference for higher allocations. Also, the last inequality says that higher type has a stronger preference for each allocation to outside option of the type. As the term implies, in a single crossing screening problem the agent's preference over allocations has the *single crossing property* (SCP).

The single crossing condition of the screening problem (Θ, X, U) facilitates solving the problem by specifying the set of binding constraints. Because it preserves the preference for a better allocation over a worse one as type goes up, if every pair of local incentive constraints hold, all the IC constraints do.³ Moreover, the condition also preserves the preference for each

³Given a total order \succeq_{Θ} over the type set Θ , local IC constraint means the IC constraint between type θ and θ' with $\theta' \succeq_{\Theta} \theta$ such that there does not exist another type between them, *i.e.*, $\theta' \succeq_{\Theta} \theta'' \succeq_{\Theta} \theta$.

allocation over the outside option as type goes up, and thus, only the IR constraint for the lowest type needs to be satisfied.

A well-known example of a single crossing screening problem is the following single good monopoly problem.

Example 1 Suppose that (Θ, X, U) is a screening problem equivalent to the single good monopoly. That is, $\Theta \subseteq \mathbb{R}_+$ is the finite set of the consumer's possible values for the good, and $X = [0,1] \times \mathbb{R}_+$ is the allocation set where $(q,p) \in X$ means a lottery with probability q for obtaining the good and the price p. The payoff functions are defined by

$$U_A((q,p),\theta) = q\theta - p \ and \ U_P((q,p),\theta) = p.$$

Also, (0,0) is the type independent outside option \emptyset . Then, the single crossing condition of (Θ, X, U) can be proven if we choose a total order on Θ as the standard order \geq on \mathbb{R} and a total order on \succeq_X defined as

$$(q,p) \succeq_X (q',p') \Leftrightarrow q \ge q'$$

where $\emptyset = (0,0)$ is the worst allocation according to \succeq_X . More specifically, it can be shown that $U_A|_{X\times\Theta}$ satisfies the single crossing property with (\succeq_X, \geq) because for $x = (q, p) \succeq_X (q', p') = x'$,

$$U_A(x,\theta) - U_A(x',\theta) = (q-q')\theta - (p-p'): nondecreasing \ in \ \theta.$$

2.4 Reducibility to single crossing screening problem

The goal of this paper is to find a sufficient condition for the type set Θ under which given screening problem (Θ, Z, U) can be reduced to a single crossing problem (Θ, X, U) for an allocation subset X of Z without any loss in the principal's maximized payoff.

Definition 2 For a compact subset $X \subseteq Z$, a screening problem (Θ, Z, U) is reducible to a single crossing problem, or single-crossing-reducible, under X if

- 1. The screening problem (Θ, X, U) is a single crossing problem, and
- 2. For every prior belief $\mu \in \Delta(\Theta)$, every optimal IC mechanism under X is also an optimal IC mechanism under Z.

The second condition says that the optimal mechanisms under Z and X are equivalent in terms of the principal's payoff, and the equivalence is "informationally robust" with respect to distribution over the type set Θ . Hence, when finding an optimal mechanism in the original problem (1), one only needs to solve (2), which is much easier to solve due to the single crossing condition.

A tradeoff occurs when finding a subset X that satisfies both conditions together. The single crossing condition requires an order on the agent's types with which for every pair of elements of X the agent's preference between them satisfies the single crossing property. Hence, a smaller X according to the set inclusion order makes it easier to satisfy the first condition. In contrast, the second condition, the equivalence between optimal IC mechanisms under Z and X, requires X to be rich enough so that the restriction on allocations does not negatively affect the principal in screening the agent.

3 First example

To fix ideas, consider the following example of product line design. A monopolist designs a menu of products characterized by two attributes $(x_1, x_2) \in [0, 1]^{2.4}$ The cost of producing the good (x_1, x_2) is given as $C(x_1, x_2) = \frac{x_1^2 + x_2^2}{2}$. There are finite N types of buyer whose utility function is symmetric and has a constant elasticity of substitution (CES). That is, each type has a utility function of product defined by

$$u(x) = v \cdot \left(\frac{x_1^r + x_2^r}{2}\right)^{\frac{1}{r}}.$$

for some (v, r) where $v \ge 0$ is the utility from the product (1, 1) and $\frac{1}{1-r} \ge 0$ is the elasticity of substitution. Also, the buyer's utility is quasilinear in money. The buyer privately knows his type while the seller only has a prior belief $\mu \in \Delta(\Theta)$ over the types of the buyer. The monopolist seeks to find the optimal selling mechanism that maximizes expected profit.

Generally, the problem is not a single crossing one. This is true even when both values of v and r are increasing according to an order over Θ . To illustrate this, consider the following three type example. In the left panel in Figure 1, the three dots show the buyer's three types with their values of r and v. Here, type θ 's value of r is $-\infty$, and thus, the type has Leontief utility function of good. Also, type θ' with r = 0 has Cobb Douglas function while type θ''

⁴If the product is a cellphone plan, one can think of x_1 as the amount of data and x_2 as the length of talk that a consumer can use.

with r = 1 has a linear utility function with perfect substitutes. The right figure shows their corresponding indifference curves over the set of products where the dots show three possible goods a, b and c.



Figure 1

First, consider the two goods a and b. Because both of them have balanced attributes (*i.e.*, $x_1 = x_2$), the size of the payoff difference from the goods is determined solely by the size of v, and thus,

$$u_{\theta''}(a) - u_{\theta''}(b) > u_{\theta'}(a) - u_{\theta'}(b) > u_{\theta}(a) - u_{\theta}(b).$$

Hence, for the single crossing property, the type order \succeq_{Θ} has to be chosen as either $\theta'' \succeq_{\Theta} \theta' \succeq_{\Theta} \theta$ or $\theta \succeq_{\Theta} \theta' \succeq_{\Theta} \theta''$.

However, if one chooses a and c, type θ'' is indifferent between them while good a is more preferred to good c by type θ and θ' . Especially, if v for type θ' is sufficiently larger than v for type θ , so type θ' has a much stronger preference for the good with better attributes among balanced goods, it can be the case that type θ' has a stronger preference for good a over c than type θ . That is,

$$u_{\theta'}(a) - u_{\theta'}(c) > u_{\theta}(a) - u_{\theta}(c) > u_{\theta''}(a) - u_{\theta''}(c).^{5}$$

Hence, for the single crossing property, the type order \succeq_{Θ} has to be chosen as either $\theta' \succeq_{\Theta} \theta \succeq_{\Theta} \theta''$ or $\theta'' \succeq_{\Theta} \theta \succeq_{\Theta} \theta'$. Thus, no matter how the goods are ordered, there does not exist a single order of the agent's types that captures the strength of their preference for higher good. This shows that the problem is not a single crossing one.

⁵For example, this holds when a = (0.6, 0.6) and c = (0.3, 0.9).

Despite not having the single crossing property, it turns out that when both v and r are increasing according to a type order, the screening problem is single-crossing-reducible under the set of goods with balanced attributes paired with price, denoted as

$$X := \{ (x, x) | x \in [0, 1] \times \mathbb{R}_+ \}.$$

That is, in this case, screening problem with allocation space restricted to X satisfies the single crossing condition and also preserves the seller's maximized profit.

First, I show that the screening problem becomes a single crossing one when the allocation set is restricted to be X. This is because v is the MRS between the balanced increase in the attributes and money at every good with balanced attributes. Hence, if the seller has to sell only goods with balanced attributes, each type's strength of preference for goods with better attributes is determined by only the value of v.⁶ Thus, the single crossing property holds when the agent with higher v is higher type and allocation that gives better attributes is higher allocation. Without loss of generality, label each type as n for some $n \in \{1, \dots, N\}$ so that higher type n has higher v. Because of the single crossing property, it is well-known that if only local IC constraints hold, all the other IC constraints also hold. However, I here use another property of the single crossing problem with the principal's type-independent preference: the solution of the single crossing problem is a solution of a more relaxed problem with fewer constraints only against *downward deviations*. That is, in the relaxed problem, the seller only needs to prevent higher type buyers from reporting a lower type, but this does not strictly increase the seller's payoff.⁷

Next, I show that the seller's maximized profit is preserved even when the allocation set is restricted to be X. Due to the equivalence between the solutions of the single crossing problem and its relaxed one in the previous paragraph, it suffices to prove that this holds in the relaxed problem where the seller only needs to prevent downward deviations.⁸ In other words, allocation outside X is not helpful for the principal who only considers downward deviations.

This holds because r is the MRS between larger attribute and the other and the value of r for type n is increasing in n, which means that for the agent with higher n the larger attribute is better substitute for the other. Graphically, this implies that in the space of goods, the indifference curves of all types of the agent are tangent with each other at every point on the

⁶That is, from each good (q, q) for $q \in [0, 1]$, the buyer's utility is given as u((q, q)) = vq, which is equivalent to the screening problem addressed by Mussa and Rosen (1978).

⁷The property will be generalized in Section 4.4 while the relaxed problem will be more formally defined in Section 4.2.1.

⁸For more details, see Figure 3 in Section 4.2



Figure 2

45 degree line. Also, a lower type agent's indifference curve is located above the indifference curve for any higher type. Moreover, by the symmetry of the production cost function, the principal's isocost curve is also tangent with the agent's indifference curves. Figure 2 illustrates the tangency between the curves.

Now, I argue that the tangency condition implies that when only downward deviations need to be prevented, the principal does not have to sell the goods without balanced attributes. Suppose that good x(n) with a price is assigned to type n. Then, if one moves the type nallocation toward 45 degree line with preserving type n agent's payoff and the price, every agent with a higher type will get a lower payoff from the type n allocation, which means that the agent's incentive for any downward deviation is weaker. Moreover, the principal can save on production cost due to its symmetric cost function.

Due to the tangency condition and its implication for the agent's downward deviation incentives, I call the 45 degree line the *incentive contract curve*.⁹ The key features and approach here will be generalized in Section 4.

4 Reducibility to single crossing problem

4.1 Statement of the main result

My main result specifies a sufficient condition for the single-crossing-reducibility of (Θ, Z, U) under X. The reduction is possible if with a total order on Θ , X is an incentive contract curve

⁹Contract curve in the Edgeworth box satisfies the similar tangency condition.

and the screening problem (Θ, X, U) is a single crossing problem and has increasing conflict of interest property. Below I define these terms and formally state the main theorem.

First, I define an incentive contract curve in the general setting. In words, an incentive contract curve for a total order on Θ is an allocation subset such that for each allocation z and each type θ , there exists another allocation z' in the subset that satisfies the two conditions: First, for the type θ , z is Pareto dominated by z'; and second, the agent with higher type than θ prefers z to z'.

Definition 3 An allocation subset $X \subseteq Z$ is an *incentive contract curve* with a total order \succeq_{Θ} on Θ if there exists $\pi : Z \times \Theta \to X$ s.t. for all $z \in Z$ and $\theta, \theta' \in \Theta$,

$$\theta' \succeq_{\Theta} \theta \implies \begin{cases} U_A(\pi(z,\theta),\theta) \ge U_A(z,\theta) \\ U_P(\pi(z,\theta),\theta) \ge U_P(z,\theta) \\ U_A(\pi(z,\theta),\theta') \le U_A(z,\theta') \end{cases}$$
(C)

Next, for the incentive contract curve X with a total order \succeq_{Θ} on Θ , the screening problem (Θ, X, U) has to not only be a single crossing one but also have an increasing conflict of interest with the same order \succeq_{Θ} . Intuitively, an increasing conflict of interest means that as type goes up the principal is more likely to find it less profitable to give the agent an allocation that is preferable to different types of agent.

Definition 4 A screening problem (Θ, X, U) has an *increasing conflict of interest* with a total order \succeq_{Θ} on Θ if for every $x, x' \in X$ and $\theta, \theta' \in \Theta$ with $\theta' \succeq_{\Theta} \theta$,

$$\begin{cases} U_A(x',\theta') \ge U_A(x,\theta') \\ U_A(x',\theta) \ge U_A(x,\theta) \\ U_P(x',\theta') \ge U_P(x,\theta') \end{cases} \implies U_P(x',\theta) \ge U_P(x,\theta). \tag{S}$$

The main result can now be stated as the following.

Theorem 1 Suppose that there exist a total order \succeq_{Θ} on Θ and an allocation subset $X \subseteq Z$ with a total order \succeq_X s.t. with $(\succeq_X, \succeq_{\Theta})$

- X is an incentive contract curve (Property (C)), and
- (Θ, X, U) is a single crossing problem and has increasing conflict of interest (Property (P) and (S)).

Then, (Θ, Z, U) is single-crossing-reducible under X.

As will be explained in Remark 2, all of the conditions in the theorem do not conflict with each other. Also, because all these conditions are about the preferences and not affected by the type distribution μ , the single-crossing-reducibility of (Θ, Z, U) is informationally robust with respect to distribution over the type set Θ .

The rest of the section is organized as follows. Section 4.2 describes our approach for proving the main result. In the following subsections, I illustrate the main factors for single-crossing-reducibility. All the formal proofs in this section are relegated to Appendix A.

4.2 Proof outline

4.2.1 Optimal lower IC mechanism

To derive the main result, I first define an *optimal lower IC (LIC) mechanism under* Z with a total order \succeq_{Θ} on Θ . Given the total order \succeq_{Θ} , the mechanism is a solution of the following relaxation of (1), which requires only IC constraints against mimicking lower types according to the order \succeq_{Θ} , so called *downward IC constraints*:

$$\max_{h:\Theta \to Z} \mathbb{E}[U_P(h(\theta), \theta)]$$
(3)
s.t. $U_A(h(\theta), \theta) \ge U_A(h(\theta'), \theta)$ for all $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$
 $U_A(h(\theta), \theta) \ge 0$ for all $\theta \in \Theta$

I call an *LIC mechanism* any direct mechanism that satisfies all the IR and downward IC constraints.

Similarly, for an allocation subset $X \subseteq Z$, I can define optimal lower IC (LIC) mechanism under X with a total order \succeq_{Θ} on Θ as the solution of the following problem:

$$\max_{h:\Theta \to X} \mathbb{E}[U_P(h(\theta), \theta)]$$
(4)

s.t. $U_A(h(\theta), \theta) \ge U_A(h(\theta'), \theta)$ for all $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$

 $U_A(h(\theta), \theta) \ge 0$ for all $\theta \in \Theta$

Again, by using the compactness argument, it can be easily shown that both optimal LIC mechanisms exist.

4.2.2 Payoff equivalence between optimal IC mechanisms

When finding a sufficient condition for the single-crossing-reducibility of (Θ, Z, U) under X, the main challenge is finding when the second condition in Definition 2, the payoff equivalence between the optimal IC mechanisms under Z and X, holds. To do so, I use the principal's ranking of the four optimal mechanisms: optimal LIC and IC mechanisms under Z and X with a total order \succeq_{Θ} on the type space Θ .



Figure 3

In Figure 3, the four preference relations show the ranking of the mechanisms based on the principal's expected payoffs. First, given the allocation set Z or X and a total order \succeq_{Θ} on Θ , the principal always gets a weakly larger expected payoff from optimal LIC mechanism than optimal IC mechanism. This is because the former, which requires only downward IC constraints, is a relaxation of the latter. Next, given the set of IC constraints that contains all the ICs or only downward ICs, optimal mechanism under Z always gives a weakly larger payoff to the principal than optimal mechanism under X. This is again because the latter under the allocation subset Z is a solution of a more relaxed one than the former under $X \subseteq Z$. More specifically, the restriction on the feasible allocations can negatively affect the principal in screening the agent.¹⁰

The preference relations in Figure 3 imply that if optimal LIC mechanism under Z with \succeq_{Θ} is equivalent to optimal IC mechanism under X, all four optimal mechanisms are equivalent. Hence, for the payoff equivalence between optimal IC mechanisms under X, it suffices to find a condition under which the following equivalences hold together:

¹⁰For example, in two good monopoly when the values are additive and perfectly negatively correlated, full surplus extraction is possible by bundling the two goods. However, if under allocation subset X bundling is not allowed, the seller, who has to sell the goods separately, cannot help paying some information rent, which is less profitable for her.

- 1. An optimal LIC mechanism under X is an optimal LIC mechanism under Z.
- 2. An optimal IC mechanism under X is an optimal LIC mechanism under X.

Remark 1 If the four optimal mechanisms are equivalent, this also implies that the four sets of optimal mechanisms have the following set inclusion relations as illustrated by Figure 4. Especially, the set of all optimal IC mechanisms under X is a subset of the set of all optimal IC mechanisms under Z.



Figure 4

In Section 4.3, I show that incentive contract curve X (Property (C)) guarantees the first equivalence, the equivalence between optimal LIC mechanisms under X and Z. In Section 4.4, I show that single crossing problem (Θ, X, U) with increasing conflict of interest (Property (P) with (S)) guarantees the second equivalence, the equivalence between optimal IC and LIC mechanism under X and Z. Then, because all the conditions are independent of the type distribution μ , the equivalence between optimal IC mechanisms under X and Z is informationally robust with respect to distribution over the type set Θ .

4.3 Equivalence between optimal LIC mechanisms under Z and X

In this subsection, I show that optimal LIC mechanisms under Z and X are equivalent if X is an incentive contract curve (Property (C)). First, recall Property (C) in Definition 3: There exists $\pi: Z \times \Theta \to X$ s.t. for all $z \in Z$ and $\theta, \theta' \in \Theta$,

$$\theta' \succeq_{\Theta} \theta \implies \begin{cases} U_A(\pi(z,\theta),\theta) \ge U_A(z,\theta) \\ U_P(\pi(z,\theta),\theta) \ge U_P(z,\theta) \\ U_A(\pi(z,\theta),\theta') \le U_A(z,\theta') \end{cases}$$
(C)

Intuitively, the property says that allocations outside the incentive contract curve are not useful for a principal who only needs to prevent the agent's downward deviations. That is, if an LIC mechanism assigns allocation z to type θ , by replacing z with $\pi(z, \theta)$ the principal's payoff can be increased without violating any downward IC constraint. Therefore, for an incentive contract curve X, optimal LIC mechanisms under Z and X are equivalent in terms of the principal's payoff.

Proposition 1 Suppose that X is an incentive contract curve with a total order \succeq_{Θ} on Θ . Then, for all type distributions $\mu \in \Delta\Theta$, with \succeq_{Θ}

- Optimal LIC mechanisms under Z and X give the principal the same expected payoff.
- The set of all optimal LIC mechanisms under X is a subset of the set of all optimal LIC mechanisms under Z.

It is worth noting that for every total order on the type set, an incentive contract curve always exists. This is because the allocation set Z itself is an incentive contract curve for π defined by $\pi(z, \theta) = z$ for every $z \in Z = X$ and $\theta \in \Theta$. However, for the single crossing condition of (Θ, X, U) , one needs to find a sufficiently small incentive contract curve according to the set inclusion order. Appendix B provides one way to construct such an incentive contract curve to implement the main result.

In the example of Section 3, it can be shown that when type n's value of r is nondecreasing in $n, X = \{(q,q) | q \in [0,1]\} \times \mathbb{R}_+$ is an incentive contract curve with the order \geq on $\Theta = \{1, 2, \dots, N\}$.¹¹ Defining $\pi : Z \times \Theta \to X$ s.t. for each n and each allocation $((q_1, q_2), p) \in Z = [0,1]^2 \times \mathbb{R}_+$,

$$\pi(((q_1,q_2),p),n)=((q_0,q_0),p)$$

where (q_0, q_0) is the good with balanced attributes that gives the same payoff to type θ as (q_1, q_2) , *i.e.*, $q_0 = u_n(q_1, q_2)$. Hence, the first inequality about payoffs in Property (C) holds.

¹¹Recall that each type is labeled as one of the numbers from 1 to N so that type n's value of v is increasing in n.

As illustrated by Figure 2, the other two inequalities hold because of the tangency condition between the indifference curves of all players in the product space.

4.4 Equivalence between optimal LIC and IC mechanisms under X

In this section, I show that in a single crossing screening problem, optimal LIC and IC mechanisms are equivalent if the problem satisfies increasing conflict of interest (Property (S)). In other words, any optimal IC mechanism is a solution of a more relaxed problem that only requires the IR and downward IC constraints. The property of increasing conflict of interest is about how the relation between principal and agent's preferences change as the type increases. First, recall Property (S) in Definition 4:

For every $x, x' \in X$ and $\theta, \theta' \in \Theta$ with $\theta' \succeq_{\Theta} \theta$,

$$\begin{cases} U_A(x',\theta') \ge U_A(x,\theta') \\ U_A(x',\theta) \ge U_A(x,\theta) \\ U_P(x',\theta') \ge U_P(x,\theta') \end{cases} \implies U_P(x',\theta) \ge U_P(x,\theta). \tag{S}$$

The property says that for any pair of types and any pair of elements of X, if both types of agent have the same preference over the allocations, and the principal with the higher type agent also shares the same preference over the allocations, then does the principal with the lower type. Roughly, for a pair of allocations that cannot be used to screen the types of agent, the principal with the lower type is more likely to have the same preference over the pair of allocations with those types of agent, which implies less conflict of interest between the principal and the agent at the lower type. Hence, the principal does not have an incentive to violate the upward IC constraint by assigning the more preferable allocation only to the higher type even when only downward IC constraints need to be satisfied.

For an intuition for the equivalence result, consider the case of binary type with binary allocation. Suppose that the property does not hold for the two allocations x and x'. That is, the principal with the low type strictly prefers x to x' while the others, both types and the principal with the high type, prefers x' to x. Then, if only downward IC mechanism needs to be satisfied, the principal optimally assigns x' to the high type and x to the low type rather than giving both types the same allocation, either x or x'. However, this does not satisfy the upward IC constraint. Because of the single crossing property of the agent's preference, this logic of binary type example can be easily extended to the general type case. The formal proof of the proposition is relegated to Appendix A.2.

Proposition 2 Suppose that (Θ, X, U) is a single crossing screening problem and has increasing conflict of interest with $(\succeq_X, \succeq_{\Theta})$. Then, for all type distributions $\mu \in \Delta\Theta$, every optimal IC mechanism under X is an optimal LIC mechanism under X with \succeq_{Θ} .

The property of increasing conflict of interest always holds if the principal's preference is type-independent. In the case of type dependent preference of the principal, the property also holds if the players' preferences U, which integrates the preferences of the principal and the agent U_P and U_A over X, has the single crossing property with an order of the players satisfying some conditions, which is addressed in Appendix C. Especially, Appendix D shows that when the players are ordered in a particular fashion, an optimal IC mechanism under X assigns the same allocation to every type of agent. This leads to the optimality of no screening in the original problem if X is an incentive contract curve. Also, in Appendix G, I show the property can be further weakened under some assumptions, such as the existence of money transfers.

Remark 2 The three conditions in the main result, Property (C) for X being an incentive contract curve, Property (P) for the single crossing property and Property (S) for increasing conflict of interest of (Θ, X, U) , do not conflict with each other. (C) is about the players' preferences for elements over non-elements of X while (P) and (S) are about their' preferences over X. Moreover, (P) is only about how type affects the agent's preference while (S) is about how type affects the relations between the principal and the agent's preferences.

5 Optimality of nested mechanism

In this section, I consider a case that is particularly easy to apply the main result. In this case, the principal and the agent have linear payoff functions, and a dominance relation, similar to a stochastic dominance, is defined over allocations. These conditions facilitate finding sufficient conditions for single-crossing-reducibility. First, the linearity of the payoff functions allows us to guess that an incentive contract curve X can be located on the boundary of the allocation set Z. Moreover, combined with the dominance relation over allocations, it enables us to find simple sufficient conditions for Property (C) with X having a desirable ordering property. By using the ordering property of X, I find conditions for Property (P) and (S), and this also leads to a *nested structure* of optimal IC mechanism. That is, all the available allocations in

the optimal mechanism are totally ordered by the standard componentwise order. Although this case may seem stylized, it will turn out that many well-known screening problems can be reduced to problems that satisfy the conditions. In the next section, I will apply the results here to various economic settings such as multiple good monopoly, persuasion and delegation. All the formal proofs are relegated to Appendix \mathbf{E} .

5.1 Additional setup

Each allocation $z := (z^+, z^-)$ consists of values for positive components $z^+ : C^+ = \{1, \dots, M\} \rightarrow \mathbb{R}_+$ and those for negative components $z^- : C^- = \{1, \dots, N\} \rightarrow \mathbb{R}_+$. For each positive component m, every type of the agent prefers larger $z^+(m)$ while for each negative component n, each type prefers smaller $z^-(n)$. Allocation $z \equiv 0$ is type-independent outside option.

Next, I define the following orders on allocations. First, for $z \in Z$, define $s_z^+(m) := \sum_{m' \le m} z^+(m'), \ s_z^-(n) := \sum_{n' \le n} z^-(n').$

Definition 5 For $k \in \{0, 1, 2\}$, the kth dominance order \succeq_k is a partial order on Z s.t.

- $(\mathbf{k} = \mathbf{0}) \ z' \succeq_0 z \Leftrightarrow For \ all \ c \in C, \ z'(c) \ge z(c)$
- $(\mathbf{k} = \mathbf{1}) \ z' \succeq_1 z \Leftrightarrow For \ all \ m, \ n, \ s_{z'}^+(m) \ge s_z^+(m), \ s_{z'}^-(n) \ge s_z^-(n).$
- $(\mathbf{k} = \mathbf{2}) \ z' \succeq_2 z \Leftrightarrow For \ all \ m, \ n,$

$$\sum_{m' \leq m} s^+_{z'}(m') \geq \sum_{m' \leq m} s^+_{z}(m'), \ \sum_{n' \leq n} s^-_{z'}(n') \geq \sum_{n' \leq n} s^-_{z}(n')$$

The 0th dominance order is the standard component-wise order. The first and second dominance orders are similar to the first and second stochastic dominance orders. Especially, for probability distributions z^+ and z'^+ over C^+ and z^- and z'^- over C^- , that z' first (second) order dominates by z means that z^+ and z^- respectively first (second) order stochastically dominate by z'^+ and z'^- .

By using the dominance orders, I define the allocation set. Let Z be a lower contour set for a fixed allocation $\bar{z} := (\bar{z}^+, \bar{z}^-)$ and an order \succeq_k for a fixed $k \in \{0, 1, 2\}$:

$$Z:=\{z=(z^+,z^-)\in\mathbb{R}^{C^+}_+\times\mathbb{R}^{C^-}_+|\bar{z}\succeq_k z\}$$

I denote $\bar{z}_0^+ := \max \operatorname{supp} \bar{z}^+$ and $\bar{z}_0^- := \max \operatorname{supp} \bar{z}^-$ which are respectively the maximum positive and negative components that \bar{z} have nonzero values.

The dominance constraint arises for different reasons in economic problems: For example, in multiple good monopoly the 0th dominance order stems from a capacity constraint related to a unit supply for each good while the first dominance order stems from a feasibility condition for the probabilistic bundling. The second dominance order is derived from the incentive compatibility constraint in a delegation problem and the constraint for the distribution of posteriors in a persuasion problem.

Lastly, I assume that the principal and the agent's payoffs are linear in allocation. Denote as $u_P^+(m,\theta)$ and $u_A^+(m,\theta)$ the principal and the agent's marginal benefits of the positive component m at θ and as $u_P^-(n,\theta)$ and $u_A^-(n,\theta)$ their marginal losses of the negative component n at θ . According to the definition of positive and negative components, $u_A^+(m,\theta)$ and $u_A^-(n,\theta)$ are nonnegative for every m, n and θ . Additionally, I assume that $u_A^-(n,\theta)$ is strictly positive for every n and θ . Then, the principal and the agent's payoff functions can be defined as

$$\begin{split} U_P(z,\theta) &= \sum_{m=1}^M z^+(m) u_P^+(m,\theta) - \sum_{n=1}^N z^-(n) u_P^-(n,\theta) = U_P^+(z,\theta) - U_P^-(z,\theta) \\ U_A(z,\theta) &= \sum_{m=1}^M z^+(m) u_A^+(m,\theta) - \sum_{n=1}^N z^-(n) u_A^-(n,\theta) = U_A^+(z,\theta) - U_A^-(z,\theta) \end{split}$$

where $U_P^+(z,\theta)$, $U_P^-(z,\theta)$, $U_A^+(z,\theta)$ and $U_A^-(z,\theta)$ are the principal and the agent's total benefits and losses from the positive and negative components, respectively.

5.2 Reduction to single crossing screening problem

5.2.1 Upper boundary set of Z

It is well-known that the contract curve in the Edgeworth box for an economy with two linear utility functions is either the right and bottom edges or the left and top edges of the box unless the utility functions are same. Likewise, due to the linearity of U, one can guess that the smallest incentive contract curve can be on the boundary of Z. Moreover, the following allocation subset, which is analogous to the two fold product of the contract curve, will be the incentive contract curve X in my analysis.

Definition 6 For $Z = \{z = (z^+, z^-) \in \mathbb{R}^{C^+}_+ \times \mathbb{R}^{C^-}_+ | \bar{z} \succeq_k z\}$, upper boundary set of Z, $\hat{bd}(Z) \subseteq Z$, is the set of all allocations such that for some $1 \leq \bar{m} \leq M$ and $1 \leq \bar{n} \leq N$

- $z^+(m) = \bar{z}^+(m)$ and $z^-(n) = \bar{z}^-(n)$ for $m < \bar{m}$ and $n < \bar{n}$
- $z^+(m) = 0 = z^-(n)$ for $m > \bar{m}$ and $n > \bar{n}$.

The definition says that each of the positive and negative components of elements of $\hat{bd}(Z)$ takes the following form: whenever the element has a strictly positive value for component k, it has the same values for all the lower components than k as the upper bound \bar{z} . This implies that the positive and negative components of allocations in $\hat{bd}(Z)$ are respectively totally ordered by the standard component wise order \geq . More specifically, for every $z, z' \in \hat{bd}(Z)$, either $z^+ \geq z'^+$ or $z'^+ \geq z^+$. In other words, either $z^+(m) \geq z'^+(m)$ for every m or $z'^+(m) \geq z^+(m)$ for every m. Likewise, for the negative components either $z^- \geq z'^-$ or $z'^- \geq z^-$.

Hence, the following order \succeq_{bd} on $\hat{bd}(Z)$ is a total order:

$$z' \succeq_{bd} z \Leftrightarrow z'^+ \ge z^+ \tag{5}$$

In the following subsections, I find the sufficient condition for single-crossing-reducibility of (Θ, Z, U) under $X = \hat{bd}(Z)$. According to Theorem 1, I need to find the conditions for the following with a total order \succeq_{Θ} on Θ :

- 1. Single crossing problem (Θ, X, U) with $(\succeq_{bd}, \succeq_{\Theta})$ (Property (P))
- 2. Increasing conflict of interest of (Θ, X, U) with \succeq_{Θ} (Property (S))
- 3. Incentive contract curve X with \succeq_{Θ} (Property (C))

5.2.2 Single crossing condition

First, I establish the single crossing condition on the upper boundary set $\hat{bd}(Z)$. The following condition, *increasing marginal rates of substitution*, means that higher type agent has a higher marginal benefit from a positive component relative to the marginal loss from a negative one.

Definition 7 A type set Θ has increasing marginal rates of substitution (MRS) with a total order \succeq_{Θ} on Θ if one of the followings holds:

- 1. $C^- \neq \emptyset$; and for every *m* and *n*, $\frac{u_A^+(m,\theta)}{u_A^-(n,\theta)}$ is nondecreasing in type.
- 2. $C^- = \emptyset$; and for every m and $\theta' \succeq_{\Theta} \theta$, $u_A^+(m, \theta') > 0$ whenever $u_A^+(m, \theta) > 0$.

Increasing MRS condition guarantees the single crossing condition of $(\Theta, \hat{bd}(Z), U)$. This is due to the totally ordered positive and negative components of allocations in the upper boundary set $\hat{bd}(Z)$. Moreover, under this condition optimal IC mechanism under the upper boundary set has a nested form, that is, all the available allocations in the optimal mechanism are totally ordered by the standard componentwise order \geq over $\mathbb{R}^{M+N}_+ \supseteq Z$. **Proposition 3** Suppose that a type set Θ has increasing MRS with \succeq_{Θ} . Then, for $X = \hat{bd}(Z)$, (Θ, X, U) is a single crossing one with $(\succeq_{bd}, \succeq_{\Theta})$. Moreover, the optimal IC mechanism under X has a nested form.

5.2.3 Sufficient condition for the increasing conflict of interest

In this section, I provide a set of sufficient conditions for increasing conflict of interest of the screening problem $(\Theta, \hat{bd}(Z), U)$. First, the property holds if the principal's payoff is independent of type, that is, for every m and n, $u_P^+(m, \theta)$ and $u_P^-(n, \theta)$ are constant for θ . For example, this holds in multiple good monopoly problems where the seller maximizes the revenue.¹²

When the principal's payoff is type dependent, a set of sufficient conditions for the increasing conflict of interest property can be easily derived by using the linearity of payoff functions. Here are some examples. First, when the agent has a positive marginal payoff from every component, the increasing conflict of interest holds if the principal's marginal benefit is nonincreasing in type.¹³ If the principal has a positive marginal payoff from every component, the property holds if the principal's preferences have the single crossing property.¹⁴

Proposition 4 Suppose that type set Θ has increasing MRS with \succeq_{Θ} . Then, for $X = \hat{bd}(Z)$, (Θ, X, U) has the increasing conflict of interest if one of the following conditions holds:

- 1. $C^- = \emptyset$; and for every m, $u_p^+(m, \theta)$ is nonincreasing in type; and $u_p^+(m, \theta) \le 0$ whenever $u_A^+(m, \theta) = 0$.
- 2. For every m, n, and θ , $u_P^+(m,\theta) > 0 > u_P^-(n,\theta)$; and for every m and $n \frac{u_P^+(m,\theta)}{u_P^-(n,\theta)}$ is nondecreasing in type.

5.2.4 Sufficient conditions for incentive contract curve $\hat{bd}(Z)$

So far, I have found conditions for the single crossing condition and the increasing conflict of interest of $(\Theta, \hat{bd}(Z), U)$ with a pair of total orders $(\succeq_{\Theta}, \succeq_{bd})$. In this section, I provide a condition that makes sure that the upper boundary set $\hat{bd}(Z)$ is an incentive contract curve

 $^{^{12}\}mathrm{See}$ Section 6.1 for this case.

 $^{^{13}}$ See Section 6.3 for this case.

¹⁴Appendix G provides a weaker version of the increasing conflict of interest, and it shows that optimal IC and LIC mechanisms are equivalent in a single crossing problem whenever the principal's marginal payoff from every component is positive. See Proposition 7 in Section 6.2 for a concrete example.

with \succeq_{Θ} . I first choose the mapping π_{bd} from Z to $X = \hat{bd}(Z)$ by

 $\pi_{bd}(z,\theta) \in \{x \in X | (U_A^+(x,\theta), U_A^-(x,\theta)) = (U_A^+(z,\theta), U_A^-(z,\theta))\}.$

Because the mapping π_{bd} has to preserve the total benefit and loss from the positive components and the negative components, it might not be well-defined. The sufficient condition for its welldefinedness can be obtained as the following due to the same logic of a well-known fact about stochastic dominance.¹⁵ Here, to strengthen the result in the case when \bar{z} does not have full support, I define that a real-valued function u over $Y \subset \mathbb{N}$ is *quasi-decreasing* with respect to cutoff $\bar{y} \in Y$ when u is nonincreasing for $y \leq \bar{y}$ and

$$y' \leq \bar{y} \leq y \implies u(y') \geq u(y).$$

Lemma 1 Suppose that $X = \hat{bd}(Z)$. Then, $\pi_{bd} : Z \times \Theta \to X$ is well-defined for each of the following three cases,

- Z with the **Oth** dominance order
- Z with the 1st dominance order and

for every
$$\theta$$
, $u_A^+(m, \theta)$ and $u_A^-(n, \theta)$ are quasi-decreasing in m and n (M₁)
with respect to \bar{z}_0^+ and \bar{z}_0^-

• Z with the 2nd dominance order and

for every
$$\theta$$
, $u_A^+(m, \theta)$ and $u_A^-(n, \theta)$ are quasi-decreasing in m and n (M₂)
with respect to \bar{z}_0^+ and \bar{z}_0^- and convex in m and n for $m \leq \bar{z}_0^+$ and $n \leq \bar{z}_0^-$.

Next, for the upper boundary set being an incentive contract curve, I find a sufficient condition for Property (C) when $\pi = \pi_{bd}$. To state the condition, the following ratios between marginal payoffs need to be defined.

Definition 8 For type θ and θ' , the ratio between marginal benefits and the ratio between marginal losses of type θ and θ' agents are defined as the following:

$$r^+_{\theta,\theta'}(m) := \frac{u^+_A(m,\theta)}{u^+_A(m,\theta')}, \ r^-_{\theta,\theta'}(n) := \frac{u^-_A(n,\theta)}{u^-_A(n,\theta')}$$

¹⁵For cumulative distribution functions F and G over a set Y, F first (second) order stochastically dominates G iff $\mathbb{E}_F[u(y)] \ge \mathbb{E}_G[u(y)]$ for all nondecreasing (nondecreasing and concave) functions u.

Similarly, for type θ , those ratios between the principal and the agent with type θ are defined

as

$$r^+_{P,\theta}(m) := \frac{u^+_P(m,\theta)}{u^+_A(m,\theta)}, \; r^-_{P,\theta}(n) := \frac{u^-_P(n,\theta)}{u^-_A(n,\theta)}$$

Property (C) holds if for every type θ and allocation z there exists a path from z to $\pi_{bd}(z,\theta)$ along which type θ 's payoff is preserved, decreases all higher types' payoffs, and increases the payoff of the principal with type θ . It can be shown that such a path exists if the marginal benefit (loss) of type θ is more (less) strongly decreasing and convex in component than that of higher type and less (more) than that of the principal with type θ in the following sense.

Proposition 5 $X = \hat{bd}(Z)$ is an incentive contract curve with a total order \succeq_{Θ} on Θ if one of the following conditions hold:¹⁶

• Z is defined with the 0th dominance order; and

for every $\theta' \succeq_{\Theta} \theta$, $r_{\theta,\theta'}^+(m)$, $-r_{\theta,\theta'}^-(n)$, $r_{P,\theta}^+(m)$ and $-r_{P,\theta}^-(n)$ are (R₁) quasi-decreasing in m and n with respect to \bar{z}_0^+ and \bar{z}_0^- .

- Z is defined with the 1st dominance order; and (M_1) and (R_1) hold
- Z is defined with the 2nd dominance order; and (M_2) holds; and

for every
$$\theta' \succeq_{\Theta} \theta$$
, $r_{\theta,\theta'}^+(m)$, $-r_{\theta,\theta'}^-(n)$, $r_{P,\theta}^+(m)$ and $-r_{P,\theta}^-(n)$ are (R₂)
quasi-decreasing in m and n with respect to \bar{z}_0^+ and \bar{z}_0^- and
convex in m and n for $m \leq \bar{z}_0^+$ and $n \leq \bar{z}_0^{-.17}$

$$\frac{u_{A}^{+}(m,\theta)}{u_{A}^{-}(m,\theta')}, \frac{\Delta u_{A}^{+}(m,\theta)}{\Delta u_{A}^{+}(m,\theta')}, \frac{u_{P}^{+}(m,\theta)}{u_{A}^{+}(m,\theta)}, \frac{\Delta u_{P}^{+}(m,\theta)}{\Delta u_{A}^{+}(m,\theta)} \text{ are nonincreasing in } m, \text{ and } (\hat{R_{2}}) = \frac{u_{A}^{-}(n,\theta')}{u_{A}^{-}(n,\theta)}, \frac{\Delta u_{A}^{-}(n,\theta)}{\Delta u_{A}^{-}(n,\theta)}, \frac{-\frac{u_{P}^{-}(n,\theta)}{u_{A}^{-}(n,\theta)}}{\Delta u_{A}^{-}(n,\theta)} \text{ are nonincreasing in } n$$

where $\Delta u_A^+(m,\theta) = u_A^+(m,\theta) - u_A^+(m+1,\theta), \ \Delta u_A^-(n,\theta) = u_A^-(n,\theta) - u_A^-(n+1,\theta), \ \Delta u_P^+(m,\theta) = u_P^+(m,\theta) - u_P^+(m+1,\theta), \ \text{and} \ \Delta u_P^-(n,\theta) = u_P^-(n,\theta) - u_P^-(n+1,\theta).$

¹⁶In fact, all of these conditions only need to hold for the pair of adjacent types θ and θ' because it suffices to show that $\pi_{bd}(z,\theta)^+$ and $\pi_{bd}(z,\theta)^-$ are nondecreasing in θ , which only needs to be checked locally. ¹⁷When \bar{z} has the full support, (R_2) can be weakened as the following: for every $\theta' \succeq_{\Theta} \theta$,

5.2.5 Sufficient conditions for single-crossing-reducibility

Combining all the conditions for the single-crossing-reducibility, we obtain the following result. Due to Proposition 3, this also implies that there is an optimal mechanism under Z that has a nested form.

Theorem 2 For $X = \hat{bd}(Z)$, suppose that with a total order \succeq_{Θ} the type set Θ has increasing MRS and increasing conflict of interest, and one of the following conditions also holds:

- Z is defined with the 0th dominance order; and (R_1) holds
- Z is defined with the 1st dominance order; and (M_1) and (R_1) hold
- Z is defined with the 2nd dominance order; and (M_2) and (R_2) hold.

Then, (Θ, Z, U) is single-crossing-reducible under X and there exists an optimal IC mechanism under Z that has a nested form.

All the results that I have derived in the section also hold even when C^+ and C^- are compact subsets of \mathbb{R} . Instead of the summations, Riemann integrals can be used to define dominance orders, and then, all the results including Theorem 2 with increasing and convex properties directly extend to the continuous case.

6 Applications

Theorem 2 in the previous section can be used to derive new sufficient conditions for the optimality of simple mechanisms in various settings: upgrade pricing in a multiple good monopoly problem and no screening with same cap in a delegation problem. Also, I can easily reproduce optimality conditions in the existing literature: pure bundling in the multiple good monopoly (Haghpanah and Hartline, 2021) and nested interval structure of acceptance sets in the persuasion problem (Guo and Shmaya, 2019).

6.1 Multiple good monopoly

Now, I first apply our result to a standard setting of multiple good monopoly. By using the equivalence result, I recover the optimality result of *pure bundling* that offers only the grand

bundle in Haghpanah and Hartline (2021). Then, I extend the result into upgrade pricing where all the available bundles are ordered by the set inclusion.¹⁸

Setup

There are two players: a buyer and a seller. The seller sells d products with zero production cost. The grand bundle $b^* = \{1, 2, \dots, d\}$ contains all the products, and the set of all bundles is $B = \{b | b \subseteq b^*\}$. I allow the seller to choose any randomized bundle to sell from ΔB . Hence, the allocation set Z is equal to the set of all the pairs of a randomized bundle and its relative price to the maximum price \bar{p} , that is,

$$Z = \Delta(B) \times [0,1].^{19}$$

If the buyer with type $\theta \in \Theta$ purchases a bundle $b \in B$ with price $\alpha \bar{p}$ for some $\alpha \in [0, 1]$, his payoff can be written as the following by using our notation:

$$U_A((b,\alpha),\theta) = v_\theta(b) - \alpha \bar{p}.$$

where $v_{\theta}: B \to \mathbb{R}_+$ is the value function of the buyer with type θ . The payoff function can be extended to every randomized bundle $a \in \Delta B$ by taking expectation over goods:

$$U_A((a,\alpha),\theta) = \mathbb{E}_a[v_\theta(b)] - \alpha \bar{p}.$$

The buyer's payoff from the outside option $(a, \alpha) = (0, 0)$ or not buying anything is normalized to zero. The seller's payoff is equal to the price if the buyer accepts the offer and zero otherwise:

$$U_P((a,\alpha),\theta) = \alpha \bar{p}.$$

Pure bundling

First, I reproduce the sufficient condition for pure bundling in Haghpanah and Hartline (2021).

To do so, I redefine the allocation set Z as a lower contour set of $\overline{z} := (\mathbb{1}_{b^*}, 1)$ with respect to the first dominance order. That is, for $C^- = \{1\}$ and $C^+ = B$ where M := |B| and b^* is its first component,

$$Z = \Delta(B) \times [0,1] = \{ z = (z^+, z^-) \in \mathbb{R}^{C^+} \times \mathbb{R}^{C^-} | \bar{z} := (e_1, \bar{p}) \succeq_1 z \}$$

 $^{^{18}}$ I use the definition of upgrade pricing in Bergemann, Bonatti, Haupt, and Smolin (2021).

¹⁹Here, it is without loss of generality to set the maximum price to be the maximum among the valuations that each type of agent can have for each bundle.

Here, e_1 is a unit vector in $[0, 1]^M$ with the first component equal to 1. All the bundles other than the grand bundle can be labeled from 2 to M arbitrarily. Note that the first dominance relation comes from the feasibility condition for probabilistic bundling.

For the allocation set Z, all the player's payoffs are linear in allocations. More precisely, type θ agent's marginal benefit from each m or a bundle labeled by m is his valuation for the bundle while the marginal loss can be set by the maximum price \bar{p} :

$$u_A^+(m,\theta) = v_\theta(m), \ u_A^-(1,\theta) = \bar{p}$$

And the principal's payoff is equal to the revenue:

$$u_P^+(m,\theta)=0,\;u_P^-(1,\theta)=-\bar{p}.$$

According to the definition of Z, the upper boundary set $\hat{bd}(Z)$ is the set of allocations that only assigns the buyer the grand bundle b^* :

$$X := bd(Z) = \{z \in Z | z^+(m) = 0 \text{ for every } 1 < m \le M\}$$

Therefore, if the screening problem (Θ, Z, U) is single-crossing-reducible under X, a solution for the problem is the optimal selling mechanism for (Θ, X, U) that is equivalent to the single good monopoly problem where the only good is the grand bundle. Because it is well known that posting a single price is optimal in the single good monopoly, this implies that pure bundling is optimal for (Θ, Z, U) . Hence, in order to find when pure bundling is optimal, it suffices to find a sufficient condition for the single-crossing-reducibility of (Θ, Z, U) under X, which is derived by Theorem 2.

Corollary 1 (Haghpanah and Hartline, 2021) Pure bundling is optimal for all distribution $\mu \in \Delta \Theta$ if for every $b \in B$, $v_{\theta}(b^*) \geq v_{\theta}(b)$ and there exists a total order on Θ such that for every θ, θ' with $\theta' \succeq_{\Theta} \theta$

$$v_{\theta'}(b^*) \ge v_{\theta}(b^*) \tag{6}$$

$$\frac{v_{\theta'}(b^*)}{v_{\theta}(b^*)} \le \frac{v_{\theta'}(b)}{v_{\theta}(b)} \tag{7}$$

According to Theorem 2, (Θ, Z, U) is single-crossing-reducible under X if the four conditions, increasing MRS, increasing conflict of interest, (M_1) , and (R_1) , hold. Increasing MRS is equivalent to (6), monotone MRS between the grand bundle value and money while the increasing conflict of interest holds because of the seller's type-independent payoff. (M_1) is equivalent to that $v_{\theta}(b^*) \ge v_{\theta}(b)$ for every $b \in B$, so called the *free disposal property*, because $u_A^+(m, \theta) = v_{\theta}(m)$. Lastly, (R_1) is equivalent to (7), monotone marginal rates of substitution between the grand bundle value and all smaller bundles.²⁰

The set of conditions for the optimality of pure bundling is equivalent to the condition found by Haghpanah and Hartline (2021). As in their papers, it can be easily shown that the condition is necessary and sufficient condition for the robust optimality of pure bundling with respect to distribution over the type set. Also, the condition can be readily extended to their more general sufficient condition, the positive correlation between the grand bundle value and relative values for smaller bundles, by using the decomposition method of Strassen (1965).

The main difference from the derivation in Haghpanah and Hartline (2021) is that our approach uses a primal approach rather than their dual approach.²¹ In their paper, the optimality result is derived by using duality to construct the Lagrangian variables that are non-zero only for the downward IC constraints and make the ironed virtual valuation nondecreasing in grand bundle value. However, this construction method cannot be directly applied to find the condition for the optimality of the menu containing multiple items. In contrast, my result allows to not only reproduce the condition for the optimality of pure bundling more easily but also prove optimality for more general form of selling mechanisms such as upgrade pricing, which will be explained below. Due to its more general form, the optimality condition holds for some additive valuation functions of the buyer, unlike the condition for pure bundling.

The ease of deriving the optimality condition compared to dual approach is mainly because our result enables us to separate the process of finding a solution from the original screening problem without single crossing property into two processes: one for showing the equivalence between the original problem and a single crossing screening problem(here, equivalent to single good monopoly) and the other for finding the solution from the reduced single crossing problem. This separation of the processes makes the result not only more applicable to various settings, but also makes it easier to find the solution by reducing the screening problem to a single crossing problem.

Upgrade pricing

²⁰More specifically, $r_{\theta,\theta'}^+(m) = \frac{u_A^+(m,\theta)}{u_A^+(m,\theta')} = \frac{v_{\theta}(m)}{v_{\theta'}(m)}$ is the ratio between type θ and θ' 's valuations for the *m*th bundle when the first bundle is the grand bundle. Hence, $r_{\theta,\theta'}^+(m)$ is quasi-decreasing in *m* with respect to $\bar{z}_0^+ = 1$ if (7) holds.

 $^{^{21}}$ Yang (2021) also uses equivalent primal approach to extends this result to the case when both bundling and quality discrimination are allowed.

Next, I restrict attention to the case when the value function v_{θ} is additive.²² Then, I apply my result to find a sufficient condition about the type set for the optimality of upgrade pricing, which is independent of distribution over the type set.

For an additive value function v_{θ} , the value of buyer with type $\theta \in \Theta$ for each randomized bundle $a \in \Delta B$ is determined by his value for each good l, $v_{\theta}(l)$, and the total expected probability with which the buyer will get good i: $v_{\theta}(a) = \sum_{l=1}^{d} v_{\theta}(l) \mathbb{E}_{a}[\mathbb{1}_{l \in b}]$. Therefore, without loss of generality the set of all possible bundle ΔB can be replaced with $[0, 1]^{d}$ whose each element $(q_{1}, q_{2}, \dots, q_{d})$ means that it gives the buyer each good l with probability q_{l} . Thus, the allocation set Z is equal to $[0, 1]^{d} \times [0, 1]$, which is the lower contour set of the constant function $\bar{z} := \mathbb{1}$ with respect to the 0th dominance order: For $C^{+} = \{1, \dots, d\}$ and $C^{-} = \{1\}$,

$$Z = [0,1]^d \times [0,1] = \{ z = (z^+, z^-) \in \mathbb{R}^{C^+} \times \mathbb{R}^{C^-} | \bar{z} := \mathbb{1} \succeq_0 z \}$$

Again, for Z all the players' payoffs are linear in allocations with the following marginal benefit and loss from each component of allocation:

$$\begin{split} u_A^+(m,\theta) &= v_\theta(m), \ u_A^-(1,\theta) = \bar{p} \\ u_P^+(m,\theta) &= 0, \quad u_P^-(1,\theta) = -\bar{p} \end{split}$$

The upper boundary set $\hat{bd}(Z)$ from Z is the set of allocations such that for some k they assign the buyer the goods with the smallest (k-1)th index for sure and good k for some probability:

$$X := bd(Z) = \{ z \in Z | \exists k \text{ s.t. } z^+(m) = 0 \text{ if } m > k \text{ and } 1 \text{ if } 1 \le m < k \}.$$

Note that all the available bundles in X are totally ordered with the order \succeq_{bd} on $\hat{bd}(Z)$ given in (5), which is the standard component-wise order on the set of bundles. Hence, according to the definition of upgrade pricing, the optimal selling mechanism under X is upgrade pricing. Therefore, the sufficient condition for the optimality of upgrade pricing can be obtained by finding when (Θ, Z, U) is single-crossing-reducible under X. Due to Theorem 2, the condition will turn out below to be type-monotonicity of valuation for each good and MRS between every pair of goods. Moreover, by using the linear payoff functions of the players with the single crossing condition of (Θ, X, U) , the same intuition for the optimality of upgrade pricing a monopoly price in the single good setting leads to the optimality of upgrade pricing with the menu of

²²In the case of additive valuation, because for every good $l \in \{1, \dots, d\}$ $v_{\theta}(l)$ is nonnegative, it always holds that for every bundle $b \in B$ $v_{\theta}(b^*) \ge v_{\theta}(b)$.

deterministic bundles from $\{[1], [2], \dots, [d]\} \subseteq X^*$ where for $1 \leq k \leq d$ $[k] := \{1, \dots, k\}$ is the bundle containing all the goods with index lower than equal to k. In addition, for each deterministic upgrade pricing, type-monotone valuations enable us to construct an equivalent separate pricing in terms of the allocation to each type and the prices for the bundles used by the upgrade pricing. Therefore, separate sales is also optimal.

Corollary 2 Upgrade pricing is optimal for all distribution $\mu \in \Delta \Theta$ if there exists a total order on Θ and a permutation σ of the set of goods $\{1, \dots, d\}$ such that for every θ, θ' with $\theta' \succeq_{\Theta} \theta$ and m, m' with $m' \ge m$,

> Monotone valuation: $v_{\theta'}(m) \ge v_{\theta}(m) \ \forall 1 \le m \le d$ Monotone MRS: $\frac{v_{\theta'}(\sigma(m'))}{v_{\theta'}(\sigma(m))} \ge \frac{v_{\theta}(\sigma(m'))}{v_{\theta}(\sigma(m))}$

Furthermore, there exists an optimal mechanism whose allocation is deterministic and belongs to $\{[1]_{\sigma}, [2]_{\sigma}, \cdots, [d]_{\sigma}\}$. Moreover, separate monopoly pricing is also optimal.²³

It is worth noting the difference between our condition and those of Bergemann, Bonatti, Haupt, and Smolin (2021)(henceforth, BBHS). First, as already emphasized in the proposition, unlike those of BBHS, my condition (*monotone valuation* and *monotone MRS*) is sufficient for the optimality of upgrade pricing irrespective of type distribution $\mu \in \Delta\Theta$. Conditions from BBHS are dependent on the type distribution μ because they require that the type distribution is "mostly regular" so that they can sequentially apply on ironing procedure for each good.²⁴ Therefore, there can exist some Θ such that my condition fails, but for some μ the conditions of BBHS hold.²⁵

However, for some Θ satisfying our condition, there can exist a type distribution that does not satisfy the regularity conditions of BBHS. The conditions of BBHS are to reduce the problem to *d* separate single good problems, which necessitates some regularity for each good. In contrast, under my condition the problem is reduced to a single crossing screening problem under the allocation subset *X*. Therefore, at least under *X*, regularity is not required for the

 $^{^{23}}$ Yang (2021) derives another sufficient condition for the optimality of upgrade pricing when the buyer's valuation function is non-additive.

²⁴Roughly, they require that candidate ironing intervals for each item k, the interval under which the monotonicity of virtual valuation is violated, are almost disjoint between different goods.

²⁵For example, in the case when the type distribution is mostly regular, BBHS requires only a weak monotonicity of valuations that requires order comparisons only between a pair of types above and below a cutoff, rather than every pair of types.

optimality of upgrade pricing if only a single crossing property holds on the buyers' preferences, which is equivalent to *monotone valuation*.

The condition is a necessary condition for the optimality of upgrade pricing in some cases. That is, when the type set has *strictly monotone valuation*, which means that valuation for each good is strictly increasing in type, monotone MRS becomes necessary and sufficient for the robust optimality of upgrade pricing with respect to type distribution.

As in the example, this can be shown by finding a type distribution for which upgrade pricing is not optimal. When valuation for each good is strictly increasing in type, for every two point distribution optimal mechanism would have the form of upgrade pricing. So, I need to find a three point distribution. There are two cases: If valuation for each good is sufficiently convex in type, I can construct a three point distribution for which optimal IC mechanism has only the local downward IC constraints binding, but it is not an upgrade pricing. If not, it has all the downward IC constraints binding, which also means that upgrade pricing is not optimal due to the strictly monotone valuation or the resulting strict single crossing property.

Proposition 6 Suppose that the type space Θ has strictly monotone valuation with a total order \succeq_{Θ} , i.e., for every $k = 1, \dots, d$, $v_{\theta}(k)$ is strictly increasing according to the type order \succeq_{Θ} . Then, upgrade pricing is **robustly** optimal with respect to type distribution iff MRS between every pair of goods is monotone according to \succeq_{Θ} .

6.2 Bayesian persuasion

I now consider the persuasion problem studied by Guo and Shmaya (2019) where the receiver has private information about the state. As in their paper, I derive a sufficient condition for the optimality of *nested interval structure of acceptance sets*: each type's acceptance set is an interval and for every two different types, one's acceptance set is a subset of the other's. Relative to Guo and Shmaya (2019), it allows for more general form of sender's payoff function, as explained below.

Setup

There are two players: the sender and the receiver. The state set S is given as a bounded interval on the real line containing 0 with Lebesgue measure. The receiver privately knows his own type $\theta \in \Theta \subseteq \mathbb{N}$ where Θ is a finite set with the minimal element $\underline{\theta}$ and the maximum $\overline{\theta}$. The full support distribution over $S \times \Theta$ is given by a density $f(s, \theta)$ that is bounded and continuous in s. At the start of the game, the sender commits to a disclosure mechanism (Y, κ, r) about Swhere κ is a Markov kernel from S to signal set Y and $r: Y \times \Theta \to \{0, 1\}$ is recommendation function where 0 means reject and 1 means accept. After observing the signal, the receiver chooses to accept or reject. At state $s \in S$, the receiver's payoff from accepting at state s is given by a bounded function $u(s): S \to \mathbb{R}$ such that u(0) = 0, nonzero otherwise and u(s) > 0if and only if s > 0. The sender's payoff from the agent's accepting at state s is also given by a bounded function $v(s): S \to \mathbb{R}$ while his payoff from rejection at each state is zero. I assume that at every state the sender has a stronger preference for accepting over rejecting than the receiver: if for some $s u(s) \ge 0$ then v(s) > 0. That is, there exists some state $s_0 \le 0$ such that v(s) > 0 if and only if $s > s_0$.²⁶ Note that in Guo and Shmaya (2019) the receiver's payoff from accepting is additionally nondecreasing in s while the principal's payoff from accepting is always nonnegative for every state, which definitely satisfies our assumption.

Nested interval structure of acceptance sets

To fit this problem into our setting, I consider a more relaxed problem. Given a disclosure mechanism (Y, κ, r) , one can calculate the acceptance probability at each state s when the agent always follows the recommendation r. It is denoted by

$$h(s,\theta) \equiv \int r(y,\theta) \kappa(s,dy)$$

where $h(\theta) \equiv h(\cdot, \theta) : S \to [0, 1]$ is called an *acceptance set*. Let Z be the set of all such acceptance sets:

$$Z=\{z=(z^+,z^-)\in\mathbb{R}^{C^+}_+\times\mathbb{R}^{C^-}_+|\mathbb{1}\succeq_0 z\}$$

where $C^+ = S \cap [0,\infty)$ and $C^- = \{-s | s \in S \cap (-\infty,0]\}.$

Then, the principal and the agent's utilities for each type θ are determined by the acceptance set $h(\theta)$. Because I assume the common prior between the players as in Guo and Shmaya (2019), the players' expected utilities are

$$\begin{split} U_A(h(\theta),\theta) &= \int_{c\in S^+} h(c,\theta) u_A^+(c,\theta) dc - \int_{c\in S^-} h(-c,\theta) u_A^-(c,\theta) dc \\ U_P(h(\theta),\theta) &= \int_{c\in S^+} h(c,\theta) u_P^+(c,\theta) dc - \int_{c\in S^-} h(-c,\theta) u_P^-(c,\theta) dc \end{split}$$

²⁶This is without loss of generality because unlike Guo and Shmaya (2019) we do not assume that the receiver's payoff u is nondecreasing in s.

where $u_A^+(c,\theta) := f(c,\theta)u(c)$ and $u_P^+(c,\theta) := f(c,\theta)v(c)$ for every $c \in C^+$ and $u_A^-(c,\theta) := -f(-c,\theta)u(-c)$ and $u_P^-(c,\theta) := -f(-c,\theta)v(-c)$ for every $c \in C^-$.²⁷

Then, upper boundary set $\hat{bd}(Z)$ of Z is the set of all intervals [-a, b] for $a, b \ge 0$, which can be also defined as

$$X = \hat{bd}(Z) = \{(\mathbb{1}_{[0,b]},\mathbb{1}_{[0,a]}) | a, b \ge 0\}.$$

Here, I will use both notations, [-a, b] and $(\mathbb{1}_{[0,b]}, \mathbb{1}_{[0,a]})$, equivalently. The corresponding order \succeq_{bd} on $\hat{bd}(Z)$ is also defined by that

$$[-a',b'] \succeq_{bd} [-a,b] \Leftrightarrow b' \ge b_{d}$$

If (Θ, X, U) is a single crossing problem with \succeq_{bd} , optimal IC mechanism h^* under X takes the form of nested structure of acceptance sets: for every θ and θ' , $h^*(\theta) \subseteq h^*(\theta)$ or vice versa. Hence, for the optimality of nested interval structure of acceptance set for (Θ, Z, U) , it suffices to show that (Θ, Z, U) is single-crossing-reducible under X.

Compared to the original problem, the screening problem (Θ, Z, U) where the sender offers an acceptance probability is more relaxed in terms of the set of feasible acceptance probability functions. More specifically, in (Θ, Z, U) when type θ agent deviates under a mechanism h, he has to choose an acceptance set $h(\theta')$ for some $\theta' \neq \theta$. However, in the original problem type θ agent can choose not only $h(\theta')$ but also every acceptance set that can be generated by choosing whether to accept or reject for each signal. For example, each type of agent can choose every acceptance set that can be gained by the set operation between the elements of $\{h(\Theta) | \theta \in \Theta\}$. However, if an optimal IC mechanism h^* under Z has a nested interval structure of acceptance sets due to the single-crossing-reducibility of (Θ, Z, U) under X, the single crossing property of the agent's preference over X guarantees that such possible deviations would also not be profitable, and thus, h^* is optimal in the original problem.

Thus, nested interval structure of acceptance sets is optimal if (Θ, Z, U) can be reduced to a single crossing problem under X. However, Theorem 2 cannot be applied in this case because the increasing MRS condition and the increasing conflict of interest condition might conflict with each other. This is due to the common prior belief between the principal and the agent.²⁸ For example, when $s_0 = -\infty$ as in Guo and Shmaya (2019), since $\frac{u_A^+(s',\theta')}{u_A^-(s,\theta)} / \frac{u_A^+(s',\theta)}{u_A^-(s,\theta)} =$

²⁷To simplify the exposition, here I multiply the players' expected utilities $\mathbb{E}_s[h(s,\theta)u(s)|\theta]$ and $\mathbb{E}_s[h(s,\theta)v(s)|\theta]$ for each type θ by the marginal probability $\int f(s,\theta)ds$. This is without loss of generality for applying my result that is robust to distribution over the type set.

 $^{^{28}}$ In more general setup that assumes heterogeneous beliefs as in Section 5.1 of Guo and Shmaya (2019), both condition can hold together.

 $\frac{u_P^+(s',\theta')}{u_P^-(s,\theta')} / \frac{u_P^+(s',\theta)}{u_P^-(s,\theta)} \text{ for every } \theta \text{ and } s \leq 0 \leq s', \text{ type-increasing } \frac{u_A^+(s',\theta)}{u_A^-(s,\theta)} \text{ (Definition 7) and type-decreasing } -\frac{u_P^+(s',\theta)}{u_P^-(s,\theta)} \text{ (Proposition 4) cannot hold together.}$

Hence, I weaken the increasing conflict of interest condition so that it still guarantees the equivalence between optimal IC and LIC mechanisms under X in the following proposition. See Appendix G for a weaker version of increasing conflict of interest in the general setting, as well as the relevant proofs.

Proposition 7 Suppose that the type set Θ has increasing MRS. Then, optimal IC and LIC mechanisms under $X = \hat{bd}(Z)$ give the principal the same expected payoff if $\frac{v(s)}{u(s)} \leq \frac{v(s')}{u(s')}$ for every $s \leq s_0 \leq 0 \leq s'$.

By using this proposition with Theorem 2, a sufficient condition for the reduction to a single crossing problem can be obtained.

Corollary 3 An optimal mechanism has a form of nested interval structure of acceptance sets if the following conditions hold together:

for every
$$s' \ge s$$
, $\frac{f(s', \theta)}{f(s, \theta)}$ is nondecreasing in type (8)

for
$$s \le s' < 0$$
 and $0 < s \le s', \quad \frac{v(s)}{u(s)} \ge \frac{v(s')}{u(s')}$ (9)

for
$$s \le s_0 \le 0 \le s', \quad \frac{v(s)}{u(s)} \le \frac{v(s')}{u(s')}.$$
 (10)

It is worth noting that if as in Guo and Shmaya (2019) the receiver's payoff u(s) is nondecreasing in s and the sender's payoff v(s) is positively constant of s (that is, $s_0 = -\infty$), our optimality result holds if only (8) holds, which is equivalent to the *i.m.l.r* assumption made in Guo and Shmaya (2019). Compared to Guo and Shmaya (2019), a stronger result about the equivalence between the optimal LIC and IC mechanisms under the allocation subset, which is Proposition 7, allows us to extend their result to a persuasion problem with more general form of the sender's payoff function.²⁹

²⁹In Guo and Shmaya (2019), the equivalence result is shown by a construction of an optimal IC mechanism from an optimal LIC mechanism through expanding acceptance sets for higher types. This process can be done with keeping the sender's expected payoff being increased only when the sender always prefers the receiver's accepting to rejecting.

6.3 Optimal Delegation

In this section, I study a delegation problem where the agent's private bias is constant and one-sided. Applying my result, I derive sufficient conditions for the optimality of no screening with offering the same cap to the agent regardless of his bias.

Setup

The principal has the legal right to make the decision, but only the expert is knowledgeable to make a right decision depending on the true state $s \in [0, 1]$. That is, the expert perfectly observes s before making an action choice while the principal only knows its prior distribution G with support [0, 1] and with density g that is absolutely continuous and strictly positive for all $s \in S = [0, 1]$.

The principal and the agent's utilities depend on the implemented decision $a \in \mathbb{R}$, the state s, and the agent's type or bias $\theta \in \Theta$ where the type set Θ is finite and only has nonnegative elements. The bias $\theta \ge 0$ is unknown to the principal who only has a prior belief μ about the type. I consider a function $u : \mathbb{R} \to \mathbb{R}$ that is symmetric, differentiable, strictly concave, and maximized at 0. The principal's payoff function is defined as

$$u_P(a,s) = u(a-s),$$

while the payoff function of the agent with type θ :

$$u^{\theta}_{A}(a,s) = u(a - \theta - s).$$

Before the state is realized, the principal selects and commits to a direct mechanism $h : \Theta \to Z$ where Z is a set of direct mechanisms played after the state realization. Each direct mechanism $z \in Z$ maps from the state set S to the set of admissible actions \mathbb{R} . Note that here I restrict our attention to the deterministic mechanism where offering a lottery over different actions is not allowed as in Alonso and Matouschek (2008). Thus, each allocation $z \in Z$ is equivalent to a delegation set from which the agent chooses his preferred one.

For each type θ , the agent θ 's payoff from a delegation set $z \in Z$, denoted as $U_A(z, \theta)$, is equal to

$$U_A(z,\theta) = \int_0^1 u_A^\theta(\tilde{a}(s,z,\theta),s)g(s)ds$$

where $\tilde{a}(s, z, \theta)$ is the agent θ 's optimal action when the state is s and the delegation set z is given. The corresponding principal's payoff from offering the delegation set z to the agent θ ,

denoted as $U_P(z,\theta)$, is equal to

$$U_P(z,\theta) = \int_0^1 u_P(\tilde{a}(s,z,\theta),s)g(s)ds.$$

Then, the problem becomes equivalent to the screening problem (Θ, Z, U) without the IR constraints.³⁰

No screening with the same cap

To apply my result in this section, I need to reformulate the allocation set and the payoff functions. First, it is without loss of generality to restrict the attention to the delegation set that contains $(-\infty, \underline{c}]$ for sufficient small $\underline{c} < 0.^{31}$ Because the support of the state set is [0, 1], for each type θ it is without loss to consider an extended state set $[\underline{c} - \theta, \infty]$.

Next, I need the following lemma about the agent's optimal action choice under a delegation set.³² I define $\tilde{a}^-(s_0, z, \theta) = \lim_{s \to s_0^-} \tilde{a}(s, z, \theta)$ and $\tilde{a}^+(s_0, z, \theta) = \lim_{s \to s_0^+} \tilde{a}(s, z, \theta)$.

Lemma 2 Suppose that a delegation set $z \subseteq \mathbb{R}$ that contains $(-\infty, \underline{c}]$ is given to an agent with bias θ . Then, the agent's optimal action $\tilde{a}(s, z, \theta) : [\underline{c} - \theta, \infty) \to \mathbb{R}$ satisfies the following condition.

- 1. $\tilde{a}(s, z, \theta)$ is nondecreasing in s.
- 2. If $\tilde{a}(s, z, \theta)$ is strictly increasing and continuous on an open interval (s_1, s_2) , then $\tilde{a}(s, z, \theta) = s + \theta$ on (s_1, s_2) .
- 3. If $\tilde{a}(s, z, \theta)$ is discontinuous at s_0 for some $\theta \in \Theta$, the discontinuity must be a jump discontinuity that satisfies

•
$$u_A^{\theta}(\tilde{a}^-(s_0, z, \theta), s_0) = u_A^{\theta}(\tilde{a}^+(s_0, z, \theta), s_0)$$
, i.e., $s_0 + \theta = \frac{\tilde{a}^-(s_0, z, \theta) + \tilde{a}^+(s_0, z, \theta)}{2}$

- $\tilde{a}(s,z,\theta) = \tilde{a}^-(s_0,z,\theta)$ for $s \in [\tilde{a}^-(s_0,z,\theta) \theta, s_0)$ and $\tilde{a}(s,z,\theta) = \tilde{a}^+(s_0,z,\theta)$ for $s \in (s_0, \tilde{a}^+(s_0,z,\theta) \theta].$
- $\bullet \ \widetilde{a}(s_0,z,\theta) \in \{\widetilde{a}^-(s_0,z,\theta), \widetilde{a}^+(s_0,z,\theta)\}.$

³⁰It is straightforward to show that my result still holds even when the IR constraints are not required.

³¹When \underline{c} is sufficiently small, all types of the agent will not change their actions under the optimal mechanism even if they are allowed to choose actions from $(-\infty, \underline{c}]$ as well as the delegation sets.

 $^{^{32}}$ I omit the proof here because it is equivalent to Proposition 1 in Melumad and Shibano (1991) or Lemma 2 in Alonso and Matouschek (2008).

4. If $\tilde{a}(s, z, \theta)$ is discontinuous at s_0 for some θ , then for each $\theta' \in \Theta$ $\tilde{a}(s, z, \theta')$ is discontinuous at $s_0 - \theta + \theta'$.

Because $\tilde{a}(s, z, \theta)$ only has countable discontinuities, without loss one can assume that it is right-continuous. When $\tilde{a}(s - \theta, z, \theta) = c + \int_{c}^{s} d\tilde{a}(c - \theta, z, \theta)$, from the above lemma the following type independence and dominance relation is directly obtained.

Proposition 8 For every delegation set z, $\tilde{a}(s, z, \theta)$ satisfies the following:

- $\tilde{a}(s-\theta, z, \theta) : [\underline{c}, \infty) \to \mathbb{R}$ is independent of θ .
- $1 \succeq_2 d\tilde{a}(s-\theta, z, \theta)$ when both are over $C^+ := [\underline{c}, \infty)$.

Due to the first condition in Proposition 8, the allocation set Z can be redefined as the set of all the derivatives of optimal action choice function with $C^+ = [\underline{c}, \infty), C^- = \emptyset$, and

$$Z = \{ d\tilde{a}(s-\theta,\tilde{z},\theta) : [\underline{c},\infty) \to \mathbb{R}_+ | (-\infty,\bar{c}] \subseteq \tilde{z} \subseteq \mathbb{R} \}.$$

Also, due to the second condition of Proposition 8, Z is a subset of $\overline{Z} := \{z : [\underline{c}, \infty) \to \mathbb{R}_+ | \mathbb{1} \succeq_2 z\}.$

For the rest of this section, I assume that the loss function u is quadratic, *i.e.*, $u(x) = -x^2$. Then, for the allocation set Z, the payoff function U can be also redefined by using integration by parts as in Alonso and Matouschek (2008):

$$\begin{split} U_A(z,\theta) &= U_A((-\infty,\underline{c}],\theta) + \int_{\underline{c}}^{\infty} u_A^+(c^+,\theta) \cdot z(c^+) dc^+ \\ U_P(z,\theta) &= U_P((-\infty,\underline{c}],\theta) + \int_{\underline{c}}^{\infty} u_P^+(c^+,\theta) \cdot z(c^+) dc^+ \end{split}$$

where the players' marginal benefits from allowing to choose c are given as the followings:

$$\begin{split} u_A^+(c^+,\theta) &= (1-G(c^+-\theta))(\mathbb{E}[\tilde{c}|\tilde{c} \ge c^+-\theta] - (c^+-\theta)) \\ u_P^+(c^+,\theta) &= (1-G(c^+-\theta))(\mathbb{E}[\tilde{c}|\tilde{c} \ge c^+-\theta] - c^+). \end{split}$$

Then, upper boundary set $\hat{bd}(\bar{Z})$ of \bar{Z} is

$$\hat{bd}(\bar{Z}) := \{\mathbb{1}_{[\underline{c},c']} | c' \ge \underline{c}\},\$$

and thus, the corresponding set of the agent's optimal action choices is the set of choice functions for every cap $(-\infty, c']$. Also, because $\hat{bd}(\bar{Z}) \subseteq Z$, $\hat{bd}(Z) = \hat{bd}(\bar{Z})$. Therefore, if the screening problem (Θ, Z, U) is single-crossing-reducible under $X := \hat{bd}(Z)$, this implies that offering each type of the agent a cap is optimal in the delegation problem (Θ, Z, U) . Moreover, because all the agents prefer higher caps and do so more strongly than the principal with any type of the agent, the optimal IC mechanism offers the same cap regardless of the agent's type. Thus, from Theorem 2, it can be shown that the increasing hazard ratio of g is a sufficient condition for no screening with the same cap.

Corollary 4 Suppose that the loss function $u(x) \equiv -x^2$. Then, offering same cutoff to every type of the agent is optimal if $\frac{g}{1-G}(s)$ is nondecreasing in s.

Remark 3 Under the uniform state distribution and the power loss function, Tanner (2018) finds another sufficient condition for no screening with the same cap, which is that the players are sufficiently risk averse. By using Theorem 1, one can not only reproduce his condition but also extend it to more general payoff functions.³³

7 Conclusion

In this paper, I provided a sufficient condition about the agent's type set under which the principal can restrict attention, without profit loss, to a subset of the allocations over which the agent's preference has the single crossing property. By using the result, I have derived a sufficient condition for the optimality of *nested mechanisms* where all the available allocations are ordered with the standard componentwise order.

The results can be applied to various screening problems with multiple allocations to find sufficient conditions for the optimality of simple mechanisms. For example, in the multiple good monopoly problem, upgrade pricing is optimal if for a total order on the types the valuation for each good is increasing and the MRS between every pair of goods is monotone. Monotone MRS also becomes a necessary condition for its optimality when the valuation for each good is strictly increasing. In the delegation problem with a one-sided and constant bias and quadratic loss function, no screening offering a single cap is optimal when the hazard rate of the state distribution is increasing.

For future research, more results about the optimality of simple mechanisms might be obtained by using my result in other screening problems such as taxation, auctions, and insurance.

³³When the state distribution g is uniform, offering same cutoff to every type of the agent is optimal if u' is concave, and $\frac{u'}{u}(x)$ is nonincreasing in x when x < 0. This condition about u is satisfied by the power loss function $u(x) = -|x|^l$ for $l \ge 2$ that is assumed in Tanner (2018). See Online Appendix.

Appendix

A Proof of the main result

A.1 Proof of Proposition 1

I prove this by showing that for every LIC mechanism h under Z with \succeq_{Θ} , there exists an LIC mechanism h_X under X with \succeq_{Θ} that gives the principal at least the same expected payoff as h does.

Consider an LIC mechanism h under Z with \succeq_{Θ} , which means that for every θ , $\theta' \in \Theta$ with $\theta' \succeq_{\Theta} \theta$,

$$U_A(h(\theta'), \theta') \ge U_A(h(\theta), \theta') \tag{11}$$

$$U_A(h(\theta), \theta) \ge 0 \tag{12}$$

Then, because of the definition of incentive contract curve, there exists a function $\pi : Z \times \Theta \to X$ s.t. for each $\theta \in \Theta$ and $h(\theta)$, $\pi(h(\theta), \theta)$ satisfies the followings:

$$U_A(\pi(h(\theta), \theta), \theta) \ge U_A(h(\theta), \theta) \tag{13}$$

$$U_P(\pi(h(\theta), \theta), \theta) \ge U_P(h(\theta), \theta) \tag{14}$$

$$U_A(\pi(h(\theta), \theta), \theta') \le U_A(h(\theta), \theta') \text{ for } \theta' \text{ s.t. } \theta' \succeq_{\Theta} \theta$$
(15)

Now, I show that the mechanism $h_X : \Theta \to X$ consisting of $h_X(\theta) := \pi(h(\theta), \theta)$ for each $\theta \in \Theta$ satisfies all the IR and downward IC constraints, and it gives at least the same payoff to the principal as h.

(a) IR

As (12) says, h satisfies all the IR constraints. According to (13), for each θ $h_X(\theta)$ gives more payoff to the agent than $h(\theta)$ does, and thus, h_X also satisfies all the IR constraints.

(b) Downward IC

This is because for all θ , θ' s.t. $\theta' \succeq_{\Theta} \theta$

$$U_A(\pi(h(\theta'), \theta'), \theta') \ge U_A(h(\theta'), \theta') \ge U_A(h(\theta), \theta') \ge U_A(\pi(h(\theta), \theta), \theta')$$

where the first, second, and third inequalities are from (13), (11) and (15), respectively.

(c) Larger payoff under h_X than under h

This is because of (14) which implies that for each θ the principal will get a higher payoff from $h_X(\theta)$ than $h(\theta)$.

A.2 Proof of Proposition 2

For the proof, I focus on optimal LIC mechanisms under X that satisfy a property called *agent-optimality* defined below. For the ease of exposition, I denote as θ_n the nth lowest type according to the total order \succeq_{Θ} for the single crossing condition of (Θ, X, U) . That is, $\Theta = \{\theta_1, \theta_2, \dots, \theta_N\}$ for some N > 0 and $\theta_N \succeq_{\Theta} \dots \succeq_{\Theta} \theta_2 \succeq_{\Theta} \theta_1$.

Definition 9 An optimal LIC mechanism h under X with \succeq_{Θ} is **agent-optimal** if there does not exist another optimal LIC mechanism h' under X with \succeq_{Θ} such that for every θ ,

$$U_A(h'(\theta), \theta) \ge U_A(h(\theta), \theta),$$

and either

- for some $\theta \in \Theta$ such that $U_A(h'(\theta), \theta) > U_A(h(\theta), \theta)$, or
- $\sum_{n=1}^{N-1} n \mathbbm{1}_{h'(\theta_n) = h'(\theta_{n+1})} > \sum_{n=1}^{N-1} n \mathbbm{1}_{h(\theta_n) = h(\theta_{n+1})}.$

The definition says that an optimal LIC mechanism h under X is agent-optimal if there does not exist another optimal LIC mechanism under X such that it is weakly preferred to hby all types of the agent and either it is strictly preferred by at least one type or the variance among the allocations to the highest type agents is smaller.³⁴

Then, the following Lemma, which shows that in a single crossing problem with the property of increasing conflict of interest every agent-optimal optimal LIC mechanism is an optimal IC mechanism, implies Proposition 2.

Lemma 3 If with $(\succeq_X, \succeq_{\Theta})$ (Θ, X, U) is a single crossing screening problem and has the property of increasing conflict of interest, every agent-optimal optimal LIC mechanism under X with \succeq_{Θ} is an optimal IC mechanism under X.

Proof. Suppose not. That is, given $(\succeq_X, \succeq_{\Theta})$ with which (Θ, X, U) is a single crossing one and the property of increasing conflict of interest holds, there exists an agent-optimal optimal LIC mechanism h under X which is not an optimal IC mechanism under X, *i.e.*, h does not satisfy an upward IC constraint for a pair of types. Moreover, because of the single crossing property of the agent's preference over X, it is without loss to assume that a local upward IC

³⁴The second condition can be ruled out if the agent's preference over X has a strict single crossing property

constraint between the adjacent types θ_n and θ_{n+1} for some $1 \le n \le N-1$ does not hold. Let n^* be the minimum among such n, which means that

$$U_A(h(\theta_{n^*+1}), \theta_{n^*}) > U_A(h(\theta_{n^*}), \theta_{n^*}),$$
(16)

and for $n < n^*$,

$$U_A(h(\theta_{n+1}), \theta_n) \le U_A(h(\theta_n), \theta_n). \tag{17}$$

Step 1 Because h is agent-optimal, for every $n < n^*$, at least one of upward or downward IC constraints between θ_{n+1} and θ_n is non-binding unless $h(\theta_{n+1}) = h(\theta_n)$, which implies that $h(\theta_{n+1}) \succeq_X h(\theta_n)$.

Proof. Suppose not. That is, for $h(\theta_{n+1}) \neq h(\theta_n)$,

$$(U_A(h(\theta_{n+1}), \theta_{n+1}), U_A(h(\theta_{n+1}), \theta_n)) = (U_A(h(\theta_n), \theta_{n+1}), U_A(h(\theta_n), \theta_n)).$$
(18)

Then, one can construct two LIC mechanisms h_1^n and h_2^n by replacing $h(\theta_n)$ with $h(\theta_{n+1})$ and vice versa, respectively:

$$\begin{split} h_1^n &\equiv (h(\theta_1), \cdots, h(\theta_{n-1}), h(\theta_n), h(\theta_n), h(\theta_{n+2}), \cdots, h(\theta_N)) \\ h_2^n &\equiv (h(\theta_1), \cdots, h(\theta_{n-1}), h(\theta_{n+1}), h(\theta_{n+1}), h(\theta_{n+2}), \cdots, h(\theta_N)). \end{split}$$

Because h is an optimal LIC mechanism, the principal has to prefer h to h_1^n and h_2^n :

$$U_P(h(\theta_{n+1}), \theta_{n+1}) \ge U_P(h(\theta_n), \theta_{n+1})$$
(19)

$$U_P(h(\theta_n), \theta_n) \ge U_P(h(\theta_{n+1}), \theta_n).$$
(20)

Then, the property of increasing conflict of interest with (18) and (19) implies that

$$U_P(h(\theta_n), \theta_n) \le U_P(h(\theta_{n+1}), \theta_n),$$

which with (20) also implies that

$$U_P(h(\theta_n),\theta_n) = U_P(h(\theta_{n+1}),\theta_n).$$

Combined with (18), this equation means that for h, $h' = h_2^n$ satisfies the second condition in Definition 9, which contradicts with the agent-optimality of h. \Box

Then, I construct two mechanisms $h_1^{n^*}$ and $h_2^{n^*}$ by replacing $h(\theta_{n^*})$ with $h(\theta_{n^*+1})$ and vice versa, respectively:

$$\begin{split} h_1^{n^*} &\equiv (h(\theta_1), \cdots, h(\theta_{n^*-1}), h(\theta_{n^*}), h(\theta_{n^*}), h(\theta_{n^*+2}), \cdots, h(\theta_N)) \\ h_2^{n^*} &\equiv (h(\theta_1), \cdots, h(\theta_{n^*-1}), h(\theta_{n^*+1}), h(\theta_{n^*+1}), h(\theta_{n^*+2}), \cdots, h(\theta_N)). \end{split}$$

Step 2 Both $h_1^{n^*}$ and $h_2^{n^*}$ are LIC mechanisms under X.

Proof. It is straightforward to show that $h_2^{n^*}$ is a LIC mechanism because of (16). For $h_1^{n^*}$, it suffices to show that for $n \leq n^*$, the agent with type θ_{n^*+1} prefers $h(\theta_{n^*})$ to $h(\theta_n)$ and the outside option:

$$U_A(h(\theta_{n^*}), \theta_{n^*+1}) \ge U_A(h(\theta_n), \theta_{n^*+1}) \ge 0. \tag{21}$$

From Step 1, $h(\theta_{n'}) \succeq_X h(\theta_{n'-1})$ for every $n' \leq n^*$. Thus, due to the single crossing property of U_A , the downward IC between type $\theta_{n'}$ and $\theta_{n'-1}$ implies that the agent with type θ_{n^*+1} also prefers $h(\theta_{n'})$ to $h(\theta_{n'-1})$ while the IR for type $\theta_{n'-1}$ implies that the agent with type θ_{n^*+1} also prefers $h(\theta_{n'-1})$ to the outside option:

$$U_A(h(\theta_{n'}), \theta_{n^*+1}) \ge U_A(h(\theta_{n'-1}), \theta_{n^*+1}) \ge 0. \tag{22}$$

Hence, by combining (22) for $n' = n + 1, \dots, n^*$, (21) can be obtained. \Box

Step 3 Because *h* is an optimal LIC mechanism under *X*, the principal weakly prefers *h* to $h_1^{n^*}$:

$$U_P(h(\theta_{n^*+1}),\theta_{n^*+1}) \geq U_P(h(\theta_{n^*}),\theta_{n^*+1}).$$

Because h is agent optimal and (16) holds, the principal strictly prefers h to $h_2^{n^*}$:

$$U_P(h(\theta_{n^*}), \theta_{n^*}) > U_P(h(\theta_{n^*+1}), \theta_{n^*}).$$

These two inequalities with (16) and the downward IC between θ_{n^*} and θ_{n^*+1}

$$U_A(h(\theta_{n^*+1}), \theta_{n^*+1}) \ge U_A(h(\theta_{n^*}), \theta_{n^*+1})$$
(23)

cannot hold together due to the property of increasing conflict of interest, which is a contradiction. \blacksquare

A.3 Proof of Theorem 1

The first condition of the reducibility to a single crossing problem under X holds (Θ, X, U) is a single crossing problem. Hence, it suffices to show the second condition, the equivalence between optimal IC mechanism under Z and X. According to Proposition 2, if (Θ, X, U) is a single crossing problem with the property of increasing conflict of interest, every optimal IC mechanism under X is an optimal LIC mechanism under X. Also, according to Proposition 1, if X is an incentive contract curve, every optimal LIC mechanism under X is an optimal LIC mechanism under Z. Therefore, every optimal IC mechanism under X is an optimal LIC mechanism under Z. Because every optimal LIC mechanism under Z that satisfies all the ICs has to be an optimal IC mechanism under Z, every optimal IC mechanism under X is an optimal IC mechanism under Z, which completes the proof.

B Implementation

The main challenge in applying my result to a screening problem is finding an incentive contract curve small enough to reduce the problem to single crossing one. In this section, I present one way to construct the smallest incentive contract curve according to set inclusion order.³⁵

We first define *incentive dominance order*, which is a strict partial order on the allocation set Z, so that the incentive contract curve is the set of *maximal allocations*, that is, allocations that do not have a higher allocation according to the order.

Definition 10 Given a total order \succeq_{Θ} on the type set Θ , an incentive dominance order $\succ_{\theta,Z}$ for each type θ is a strict partial order on the allocation set Z that satisfies the following: For a pair of allocations $z, z' \in Z$, if $z \succ_{\theta,Z} z'$, the following inequalities about the players' payoffs hold with at least one strict inequality

• For type θ , z Pareto dominates by z':

$$U_A(z,\theta) \geq U_A(z',\theta), \ \ U_P(z,\theta) \geq U_P(z',\theta)$$

• All the types strictly higher than θ prefers z to z':

$$U_A(z,\theta') \leq U_A(z',\theta') \ \forall \theta' \succ_\Theta \theta.^{36}$$

³⁵Of course, because the set inclusion order is a partial one, there can be multiple number of the smallest incentive contract curve.

 $^{{}^{36}\}theta' \succ_{\Theta} \theta$ means that $\theta' \succeq_{\Theta} \theta$ but $\theta' \neq \theta$.

We call the set of maximal allocations according to $\succ_{\theta,Z}$ the **incentive frontier for** θ , denoted as Z^*_{θ} . Also, the union of the incentive frontiers for all types, $Z^*_{\succeq_{\Theta}} := \bigcup_{\theta \in \Theta} Z^*_{\theta}$, is called the **incentive frontier**.

In other words, $\succ_{\theta,Z}$ is the strict componentwise order when each allocation z is represented by a vector consisting of the players' payoffs from z and type θ , $(U_A(z,\theta), U_P(z,\theta))$, and the payoffs of the higher types of the agent from z, $\{-U_A(z,\theta')|\theta' \succ_{\Theta} \theta\}$. Because the incentive frontier Z_{θ}^* for θ is the set of maximal allocations according to the order, for every allocation $z \in Z$ and every type $\theta \in \Theta$, there exists an allocation $z' \in Z_{\theta}^*$ such that all the three conditions in the definition of incentive contract curve are satisfied when $\pi(z, \theta) = z'$. Hence, the *incentive frontier*, the union of the incentive frontiers for all types, is an incentive contract curve according to Definition 3.

Lemma 4 Given a total order \succeq_{Θ} on the type set Θ , the incentive frontier $Z^*_{\succeq_{\Theta}}$ is an incentive contract curve.

Proposition 9 Every optimal downward IC mechanism under the incentive frontier $Z^*_{\geq_{\Theta}}$ is an optimal downward IC mechanism under Z.

Hence, we obtain the following result that directly comes from our main result.

Theorem 3 Suppose that there exist a total order \succeq_{Θ} on Θ such that for a total order \succeq^* on the incentive frontier $Z^*_{\succeq_{\Theta}}$, $(\Theta, Z^*_{\succeq_{\Theta}}, U)$ is a single crossing problem with increasing conflict of interest (Property (S). Then, for every type distribution $\mu \in \Delta\Theta$, every optimal IC mechanism under $Z^*_{\succeq_{\Theta}}$ is an optimal IC mechanism under Z.

C The set of sufficient conditions for increasing conflict of interest

In this section, I provide a set of sufficient conditions for the property of increasing conflict of interest. The first sufficient condition is that the principal's preference over the allocation subset X is independent of the agent's type, *i.e.*, for every $x \in X$, $U_P(x, \theta)$ is constant for θ . An example that satisfies this condition is the standard single good monopoly where the principal maximizes the expected revenue. Combined with the single crossing condition shown in Example 1, Proposition 2 leads to the equivalence between optimal LIC and IC mechanisms in the single good monopoly. The next sufficient condition is that the single crossing property of the agent' preferences U_A over X can be extended to that of the principal's preference U_P with satisfying some properties below. First, recall the definition of the single crossing property, which is introduced by Milgrom and Shannon (1994).

Definition 11 (*Milgrom and Shannon, 1994*) Suppose that Y and Z are partially ordered set with \succeq_Y and \succeq_Z , respectively. Then, $f: Y \times Z \to \mathbb{R}$ satisfies the single crossing property in (y, z) on $\tilde{Y} \times \tilde{Z} \subseteq Y \times Z$ with (\succeq_Y, \succeq_Z) if for $y' \succeq_Y y''$ in \tilde{Y} and $z' \succeq_Z z''$ in \tilde{Z} ,

$$\begin{split} f(y',z'') &> f(y'',z'') \Rightarrow f(y',z') > f(y'',z') \\ f(y',z'') &\geq f(y'',z'') \Rightarrow f(y',z') \geq f(y'',z') \end{split}$$

In order to formalize the single crossing property of the players' preferences, we integrate $U_P|_{X\times\Theta}$ and $U_A|_{X\times\Theta}$ into $U|_{X\times\Theta}: X\times\Theta\times\{A,P\}\to\mathbb{R}$. Also, if the principal faces the agent with type $\theta\in\Theta$ denoted by $(\theta; A)$, then I say that the principal also has type θ and denote her by $(\theta; P)$. Of course, unlike the agent the principal does not know her type.

For defining the single crossing property of $U|_{X\times\Theta}$, one needs to choose orders on X and $\Theta \times \{A, P\}$. First, for the order on X, I keep using the total order \succeq_X on X such that the agent's preference $U_A|_{X\times\Theta}$ satisfies the single crossing property with $(\succeq_X, \succeq_{\Theta})$ as in Section 2.3. For an order $\succeq_{\Theta \times \{A, P\}}$ of the players' types $\Theta \times \{A, P\}$, I require it to satisfy the following two conditions together:

 (O_1) For every $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$,

$$(\theta; A) \succeq_{\Theta \times \{A, P\}} (\theta'; A).$$

 (O_2) For every $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta', (\theta'; P)$ is between two of $(\theta; A), (\theta; P)$, and $(\theta'; A)$.

The single crossing property of $U|_{X\times\Theta}$ with $(\succeq_X, \succeq_{\Theta\times\{A,P\}})$ becomes consistent with that of $U_A|_{X\times\Theta}$ with $(\succeq_X, \succeq_{\Theta})$ due to (O_1) while it implies the property of increasing conflict of interest due to (O_2) . Hence, from Proposition 2, I obtain the following corollary.

Corollary 5 Suppose that a screening problem (Θ, X, U) is a single crossing one with a pair of total orders $(\succeq_X, \succeq_{\Theta})$ on X and Θ and one of the followings holds:

 (S_1) The principal's preference U_P over X is independent of the agent's type, i.e., for every $x \in X$ and $\theta, \theta' \in \Theta$,

$$U_P(x,\theta) = U_P(x,\theta').$$

 (S_2) For every $x, x' \in X$, $U|_{X \times \Theta} : X \times \Theta \times \{P, A\} \to \mathbb{R}$ satisfies the single crossing property on $\{x, x'\} \times \Theta \times \{P, A\}$ with (\succeq_X, \succeq) for some order $\succeq_{\Theta \times \{A, P\}}$ on the set of the players' types $\Theta \times \{P, A\}$ with the properties (O_1) and (O_2) .

Then, an optimal LIC mechanism under X is an optimal IC mechanism under X for all type distributions $\mu \in \Delta \Theta$.

The following two total orders on the set of players' types $\Theta \times \{P, A\}$, \succeq^A and \succeq^D , are examples that satisfy the property (O_1) and (O_2) . Hence, (S_2) holds if $U|_{X\times\Theta}$ satisfies the single crossing property with a pair of \succeq_X and either of them.

$$\succeq_{\Theta}^{A:} \left\{ \begin{array}{l} (\theta;i) \succeq_{\Theta}^{A} (\theta';i) \text{ for } i \in \{P,A\} \text{ and } \theta, \ \theta' \in \Theta \text{ with } \theta \succeq_{\Theta} \theta' \\ (\theta;P) \succeq_{\Theta}^{A} (\theta';A) \text{ for every } \theta, \ \theta' \in \Theta \end{array} \right\}$$
$$\succeq_{\Theta}^{D:} \left\{ \begin{array}{l} (\theta;A) \succeq_{\Theta}^{D} (\theta';A) \text{ for every } \theta, \ \theta' \in \Theta \text{ with } \theta \succeq_{\Theta} n' \\ (\theta';P) \succeq_{\Theta}^{D} (\theta;P) \text{ for every } \theta, \ \theta' \in \Theta \text{ with } \theta \succeq_{\Theta} n' \\ (\theta;A) \succeq_{\Theta}^{D} (\theta';P) \text{ for every } \theta, \ \theta' \in \Theta \end{array} \right\}$$

It can be easily shown that both orders satisfy (O_1) and (O_2) . Also, under \succeq_{Θ}^A every type of the principal is in a higher rank than every type of the agent while under \succeq_{Θ}^D every type of the principal is in a lower rank than every type of the agent. In addition, \succeq_{Θ}^A arranges the types of the principal according to the order \succeq_{Θ} while \succeq_{Θ}^D arranges the types of the principal in its reverse order.

D The optimality of no screening

In this section, I derive a sufficient conditions for the optimality of no screening, *i.e.*, the principal offers every type of the agent same allocation. From Corollary 5, the following sufficient condition (\hat{S}_2) , which is stronger than (S_2) , guarantees the equivalence between optimal IC and LIC mechanisms under X. Moreover, an optimal IC mechanism under X gives the same allocation to every type of the agent, which means the optimality of no screening under X.

Proposition 10 Suppose that (Θ, X, U) is a single crossing screening problem with a pair of total orders $(\succeq_X, \succeq_{\Theta})$ on X and Θ with the following property (\hat{S}_2) :

 $(\hat{S}_2) \ U|_{X \times \Theta} : X \times \Theta \times \{P, A\} \to \mathbb{R}$ satisfies the single crossing property with $(\succeq_X, \succeq_{\Theta}^D)$ where the order \succeq_{Θ}^D on $\Theta \times \{P, A\}$ is given in Appendix C.

Then, for all type distributions $\mu \in \Delta \Theta$, every optimal IC mechanism under X is an optimal LIC mechanism under X. Furthermore, there exists an optimal IC mechanism h^* under X such that h^* is constant for θ .

Proof. For the ease of exposition, I denote $\Theta = \{\theta_1, \theta_2, \dots, \theta_N\}$ for some N > 0 and $\theta_N \succeq_{\Theta} \dots \succeq_{\Theta} \theta_2 \succeq_{\Theta} \theta_1$. The first part comes directly from Corollary 10. For the second part, suppose that given a single crossing problem (Θ, X, U) with $(\succeq_X, \succeq_{\Theta})$ and Property (\hat{S}_2) , every optimal IC mechanism under X is not constant for θ . That is, letting n^* be the maximum among n that satisfies $h(\theta_{n+1}) \neq h(\theta_n) = \dots = h(\theta_1)$ for an optimal IC mechanism h under X, n^* is strictly smaller than N. Denote as h^* the corresponding optimal IC mechanism under X for n^* .

First, because h^* is an optimal IC mechanism under X, the upward and downward IC constraints between θ_{n^*+1} and θ_{n^*} :

$$U_A(h^*(\theta_{n^*+1}), \theta_{n^*+1}) \ge U_A(h^*(\theta_{n^*}), \theta_{n^*+1})$$
(24)

$$U_A(h^*(\theta_{n^*}), \theta_{n^*}) \ge U_A(h^*(\theta_{n^*+1}), \theta_{n^*})$$
(25)

Also, from Corollary 5, h^* is also an optimal LIC mechanism under X. Using these facts, I proceed the following steps.

Step 1. The principals with type θ_{n^*+1} and θ_{n^*} prefers their own allocation to the other:

$$U_P(h^*(\theta_{n^*+1}), \theta_{n^*+1}) \ge U_P(h^*(\theta_{n^*}), \theta_{n^*+1})$$
(26)

$$U_P(h^*(\theta_{n^*}), \theta_{n^*}) \ge U_P(h^*(\theta_{n^*+1}), \theta_{n^*})$$
(27)

Proof. Suppose not. First, suppose that $U_P(h^*(\theta_{n+1}), \theta_{n+1}) < U_P(h^*(\theta_n), \theta_{n+1})$. Then, due to the single crossing property of the agent's preference over X, the following mechanism

$$h_1^{n^*} \equiv (h^*(\theta_1), \cdots, h^*(\theta_{n^*-1}), h^*(\theta_{n^*}), h^*(\theta_{n^*}), h^*(\theta_{n^*+2}), \cdots, h^*(\theta_N))$$

satisfies all the downward IC and IR constraints as well as gives the principal a strictly higher payoff than h, which is a contradiction.

Next, suppose $U_P(h^*(\theta_{n^*}), \theta_{n^*}) < U_P(h^*(\theta_{n^*+1}), \theta_{n^*})$. Then, due to the single crossing property (\hat{S}_2) , this with (24) implies that

$$U_A(h^*(\theta_{n^*}), \theta_{n^*}) \le U_A(h^*(\theta_{n^*+1}), \theta_{n^*}).$$

Hence, with (25), $U_A(h^*(\theta_{n^*}), \theta_{n^*}) = U_A(h^*(\theta_{n^*+1}), \theta_{n^*})$. Therefore, the following mechanism

$$h_2^{n^*} \equiv (h^*(\theta_1), \cdots, h^*(\theta_{n^*-1}), h^*(\theta_{n^*+1}), h^*(\theta_{n^*+1}), h^*(\theta_{n^*+2}), \cdots, h^*(\theta_N))$$

satisfies all the downward IC and IR constraints as well as gives the principal a strictly higher payoff than h^* , which is a contradiction.

Step 2. For every $n \leq n^*$, the principal with type θ_n and type θ_n agent are indifferent between $h^*(\theta_{n^*})$ and $h^*(\theta_{n^*+1})$:

$$\begin{split} U_P(h^*(\theta_{n^*}),\theta_n) &= U_P(h^*(\theta_{n^*+1}),\theta_n) \\ U_A(h^*(\theta_{n^*}),\theta_n) &= U_A(h^*(\theta_{n^*+1}),\theta_n). \end{split}$$

Proof. Due to the single crossing property (\hat{S}_2) , for every $n \leq n^*$ (24) and (26) imply

$$\begin{split} U_P(h^*(\theta_{n^*}), \theta_n) &\leq U_P(h^*(\theta_{n^*+1}), \theta_n) \\ U_A(h^*(\theta_{n^*}), \theta_n) &\leq U_A(h^*(\theta_{n^*+1}), \theta_n) \end{split}$$

while (25) and (27) imply

$$\begin{split} &U_P(h^*(\theta_{n^*}),\theta_n) \geq U_P(h^*(\theta_{n^*+1}),\theta_n) \\ &U_A(h^*(\theta_{n^*}),\theta_n) \geq U_A(h^*(\theta_{n^*+1}),\theta_n). \end{split}$$

Therefore, the above indifference relations are obtained. \Box

Hence, even after replacing $h^*(\theta_n) (= h^*(\theta_{n^*}))$ for every $n \leq n^*$ with $h^*(\theta_{n^*+1})$, the principal's payoff will be preserved without violating any IC constraints, which means that it is also an optimal IC mechanism under X. Therefore, this contradicts with the definition of n^* .

Then, a sufficient condition for the optimality of no screening under Z can be obtained by replacing the property of increasing conflict of interest in Theorem 1 with the condition (\hat{S}_2) .

Corollary 6 Suppose that there exist a total order \succeq_{Θ} on Θ and an allocation subset $X \subseteq Z$ with a total order \succeq_X s.t. with $(\succeq_X, \succeq_{\Theta})$

- X is an incentive contract curve, and
- (Θ, X, U) is a single crossing problem with property (\hat{S}_2) .

Then, for all type distributions $\mu \in \Delta \Theta$ there exists an optimal IC mechanism under Z, $h^* : \Theta \to X$ such that h^* is constant for θ .

E Proofs for the results in Section 5

E.1 Proof of Proposition 3

For the first part, it suffices to show that increasing MRS implies that for every $z' \succeq_{bd} z$ and $\theta' \succeq_{\Theta} \theta$,

$$U_A(z',\theta) \ge (>)U_A(z,\theta) \implies U_A(z',\theta') \ge (>)U_A(z,\theta')$$
(28)

$$U_A(z,\theta) \ge 0 \implies U_A(z,\theta') \ge 0 \tag{29}$$

 $z' \succeq_{bd} z$ implies that $z'^+ \ge z^+$. Because it is trivial when $C^- = \emptyset$, I focus on the case when $C^- \ne \emptyset$.

(Case 1) $z^- > z'^-$

In this case, the first condition holds because both types of the agent have the same weak or strict preference for z over z'.

(Case 2) $z'^- \ge z^-$

First, increasing MRS implies the following:

$$\min_{1 \le m \le M} \frac{u_A^+(m, \theta')}{u_A^+(m, \theta)} \ge \max_{1 \le n \le N} \frac{u_A^-(n, \theta')}{u_A^-(n, \theta)}.$$
(30)

Also, note that $z'^+ \ge z^+$ and $z'^- \ge z^-$, *i.e*, for every m and n, $z'^+(m) \ge z^+(m)$ and $z'^-(n) \ge z^-(n)$. Hence,

$$U_{A}^{+}(z',\theta) - U_{A}^{+}(z,\theta) \ge 0 \text{ and } U_{A}^{-}(z',\theta) - U_{A}^{-}(z,\theta) \ge 0.$$
(31)

$$\begin{split} & U_A(z',\theta') - U_A(z,\theta') \\ = & (U_A^+(z',\theta') - U_A^+(z,\theta')) - (U_A^-(z',\theta') - U_A^-(z,\theta')) \\ \geq & \min_{1 \le m \le M} \frac{u_A^+(m,\theta')}{u_A^+(m,\theta)} \cdot (U_A^+(z',\theta) - U_A^+(z,\theta)) - \max_{1 \le n \le N} \frac{u_A^-(n,\theta')}{u_A^-(n,\theta)} \cdot (U_A^-(z',\theta) - U_A^-(z,\theta)) \\ \geq & \max_{1 \le n \le N} \frac{u_A^-(n,\theta')}{u_A^-(n,\theta)} \cdot \left[(U_A^+(z',\theta) - U_A^+(z,\theta)) - (U_A^-(z',\theta) - U_A^-(z,\theta)) \right] \\ = & \max_{1 \le n \le N} \frac{u_A^-(n,\theta')}{u_A^-(n,\theta)} \cdot (U_A(z',\theta) - U_A(z,\theta)) \end{split}$$

The first inequality holds because z' is component-wisely greater than z. The second one comes from (30) and (31). Therefore, we can obtain (28). Due to the same logic, (29) can be proved.

The second part comes directly from the single crossing property. More specifically, allocations given to two types of the agent by an IC mechanism have to be ordered by the standard componentwise order because if not, both types have the same strict preference over the allocations, and thus, IC constraints cannot hold.

E.2 Proof of Proposition 4

1. $C^- = \emptyset$; and for every m, $u_p^+(m, \theta)$ is nonincreasing in type; and $u_p^+(m, \theta) \le 0$ whenever $u_A^+(m, \theta) = 0$.

First note that when $C^- = \emptyset$, all the allocations are totally ordered with the standard component wise order \geq , which is also equivalent to \succeq_{bd} . Now, consider the following two cases and show that the increasing conflict of interest holds:

• $x' \ge x$

In this case, $U_P(x', \theta') \ge U_P(x, \theta')$, and thus, this implies that $U_P(x', \theta) \ge U_P(x, \theta)$ because

$$U_P(x',\theta') - U_P(x,\theta') \ge U_P(x',\theta) - U_P(x,\theta).$$

due to type-increasing $u_P^+(m, \theta)$. Thus, the increasing conflict of interest holds for such x and x'.

• x > x'

In this case, $U_A(x',\theta) \ge U_A(x,\theta)$ only if for every m such that x(m) - x'(m) < 0, $u_A^+(m,\theta) = 0$, which also implies $u_P^+(m,\theta) \le 0$. Hence, $U_P(x,\theta) - U_P(x',\theta) \le 0 = U_A(x,\theta) - U_A(x',\theta)$, and thus, the increasing conflict of interest property holds.

- 2. For every $m, n, \text{ and } \theta, u_P^+(m, \theta) > 0 > u_P^-(n, \theta)$ and $\frac{u_P^+(m, \theta)}{u_P^-(n, \theta)}$ is nondecreasing in type.
 - $x^+ \ge x'^+$ and $x'^- > x^ U_A(x', \theta) < U_A(x, \theta)$, and thus, the assumption of increasing conflict of interest does not hold.
 - $x^+ \ge x'^+$ and $x^- \ge x'^ U_P(x', \theta') < U_P(x, \theta')$ unless x = x', and thus, again the assumption of increasing conflict of interest does not hold.
 - $(x'^+, x'^-) \ge (x^+, x^-)$, $U_P(x', \theta) \ge U_P(x, \theta)$ always holds because $u_P^+(m, \theta) > 0 > u_P^-(n, \theta)$ for every mand n. Thus, increasing conflict of interest holds.
 - $x'^+ > x^+$ and $x^- > x'^-$ Because $\frac{u_P^+(m,\theta)}{u_P^-(n,\theta)}$ is nondecreasing in type, the principal's preference has the single crossing property, and thus, $U_P(x',\theta') \ge U_P(x,\theta')$ implies $U_P(x',\theta) \ge U_P(x,\theta)$. Hence, increasing conflict of interest holds.

E.3 Proof of Lemma 1

Well-definedness of π_{bd}

Because of the continuity of U_A , it suffices to show that

$$\left(U^+_A(\bar{z},\theta),U^-_A(\bar{z},\theta)\right) \geq \left(U^+_A(z,\theta),U^-_A(z,\theta)\right) \quad \forall (z,\theta) \in Z\times \Theta.$$

Also, because of the symmetry in the structures for positive and negative components, I show only that

$$U^+_A(\bar{z},\theta) \geq U^+_A(z,\theta) \quad \forall (z,\theta) \in Z \times \Theta.$$

For Z with the 0th dominance order, it is evident because \bar{z} is component-wisely greater than z and each type agent's marginal benefit from each positive component is nonnegative.

For the next two cases of Z with the kth dominance order for k = 1, 2, the result directly comes from the same logic of a well-known fact about the stochastic dominance in footnote 15 because $U_A^+(z, \theta) = \sum_{m=1}^M z^+(m) u_A^+(m, \theta)$

E.4 Proof of Proposition 5

I define the payoff vectors of the principal and the agent which consist of their payoffs from each component of allocation. For each allocation z and type θ , the agent's payoff vectors respectively from the positive and negative components, $u_A^+(z,\theta) : C^+ \to \mathbb{R}_+$ and $u_A^-(z,\theta) : C^- \to \mathbb{R}_+$, are defined as $u_A^+(z,\theta)(m) \equiv z^+(m)u_A^+(m,\theta)$ and $u_A^-(z,\theta)(n) \equiv z^-(n)u_A^-(n,\theta)$ for m and n. Similarly, the principal's payoff vectors for z and θ , $u_P^+(z,\theta) : C^+ \to \mathbb{R}_+$ and $u_P^-(z,\theta) : C^- \to \mathbb{R}_+$, are also defined as $u_P^+(z,\theta)(m) \equiv z^+(m)u_P^+(m,\theta)$ and $u_P^-(z,\theta)(n) \equiv z^-(n)u_P^-(n,\theta)$ for m and n.

Stochastic dominance between payoff vectors

I first show that under the same conditions in Lemma 1, π_{bd} moves each allocation in a direction of decreasing its payoff vector in a stochastic dominance sense. Because of the symmetry in the structures for positive and negative components, I show only the stochastic dominance relation for the positive components. Let's first see

$$u^+_A(\pi_{bd}(z,\theta),\theta) \succeq_1 u^+_A(z,\theta), \; u^-_A(\pi_{bd}(z,\theta),\theta) \succeq_1 u^-_A(z,\theta)$$

in the following two cases:

- Z with the **0th** dominance order
- Z with the **1st** dominance order and (M_1)

Because the mapping $\pi_{bd}(\cdot, \theta)$ preserves type θ 's total payoff from the positive components, it suffices to show that for every m

$$\sum_{m' \le m} u_A^+(\pi_{bd}(z,\theta),\theta)(m) \ge \sum_{m' \le m} u_A^+(z,\theta)(m)$$
(32)

where $u_A^+(z,\theta)(m) \equiv z^+(m)u_A^+(m,\theta).$

First, if $m \ge \pi_{bd}(z,\theta)_0^+$, (32) holds because

$$\sum_{m' \le m} u_A^+(\pi_{bd}(z,\theta),\theta)(m) = U_A^+(z,\theta) \ge \sum_{m' \le m} u_A^+(z,\theta)(m).$$
(33)

When $m < \pi_{bd}(z,\theta)_0^+$, note that for every $m' \leq m$, $\pi_{bd}(z,\theta)^+(m') = \bar{z}^+(m')$. Thus, when Z is defined with the 0th dominance order,

$$\pi_{bd}(z,\theta)^+(m')\geq \bar{z}^+(m'),$$

and when Z is defined with the 1st dominance order,

$$\sum_{m' \le m} \pi_{bd}(z, \theta)^+(m') \ge \sum_{m' \le m} \bar{z}^+(m').$$

Hence, for Z with the 0th dominance order, (32) holds because $u_A^+(m,\theta) \ge 0$ for every m.

Also, for Z with the 1st dominance order, (32) holds because $u_A^+(m,\theta)$ is nonincreasing in $m \leq \pi_{bd}(z,\theta)_0^+$.

Next, in the case when Z is defined with the **2nd** dominance order and (M_2) holds, let's show that

$$u_A^+(\pi_{bd}(z,\theta),\theta) \succeq_2 u_A^+(z,\theta), \ u_A^-(\pi_{bd}(z,\theta),\theta) \succeq_2 u_A^-(z,\theta).$$

In this case, it suffices to show that for every m

$$\sum_{m' \le m} (m' - m) u_A^+(\pi_{bd}(z, \theta), \theta)(m) \ge \sum_{m' \le m} (m' - m) u_A^+(z, \theta)(m).$$
(34)

Again, because of (33), I only need to show this for $m < \pi_{bd}(z,\theta)_0^+$, which implies for every $m' \leq m, \pi_{bd}(z,\theta)^+(m') = \bar{z}^+(m')$. Thus, for $m < \pi_{bd}(z,\theta)_0^+$ and Z with the 2nd dominance order,

$$\sum_{m' \le m} (m' - m) \pi_{bd}(z, \theta)^+(m') \ge \sum_{m' \le m} (m' - m) \bar{z}^+(m').$$

By using the same logic of footnote 15, this implies that for a nonnegative valued, nonincreasing and convex function $g : \{1, \dots, M\} \to \mathbb{R}_+$ with g(m') = 0 for $m' \ge \pi_{bd}(z, \theta)^+$,

$$\sum_{m'\leq m}g(m')\pi_{bd}(z,\theta)^+(m')\geq \sum_{m'\leq m}g(m')\bar{z}^+(m').$$

Moreover, $u_A^+(m,\theta)$ is nonincreasing and convex in $m \leq \pi_{bd}(z,\theta)_0^+$, and thus, this implies that $g(m') := (m'-m)u_A^+(m',\theta)|_{m'\geq m}$ is nonincreasing and convex in m' with g(m') = 0 for $m' \geq \pi_{bd}(z,\theta)_0^+$. Therefore, (34) holds for $m < \pi_{bd}(z,\theta)_0^+$.

Proof of Proposition 5

 $X \subseteq Z$ is an incentive contract curve with a total order \succeq_{Θ} on Θ and $\pi_{bd} : Z \times \Theta \to X$ if for all $z \in Z$ and $\theta, \theta' \in \Theta$ with $\theta' \succeq_{\Theta} \theta$

$$\mathrm{i} \ \begin{cases} U^+_A(\pi_{bd}(z,\theta),\theta') \leq U^+_A(z,\theta') \\ U^-_A(\pi_{bd}(z,\theta),\theta') \geq U^-_A(z,\theta') \end{cases}$$

$$\text{ii } \begin{cases} U_P^+(\pi_{bd}(z,\theta),\theta) \geq U_P^+(z,\theta) \\ U_P^-(\pi_{bd}(z,\theta),\theta) \leq U_P^-(z,\theta) \end{cases} \end{cases}$$

Each of these implies $U_A(\pi_{bd}(z,\theta),\theta') \leq U_A(z,\theta')$ and $U_P(\pi_{bd}(z,\theta),\theta) \geq U_P(z,\theta)$, respectively.

Due to the same logic of footnote 15, because $U_A^+(z,\theta) = u_A^+(z,\theta') \cdot r_{\theta,\theta'}^+$, if $r_{\theta,\theta'}^+(m)$ is quasi-decreasing in m with respect to \bar{z}_0^+ ,

$$u_A^+(\pi_{bd}(z,\theta'),\theta') \succeq_1 u_A^+(z,\theta') \implies U_A^+(\pi_{bd}(z,\theta'),\theta) \ge U_A^+(z,\theta) = U_A^+(\pi_{bd}(z,\theta),\theta)$$

Hence, $\pi_{bd}(z, \theta')^+ \ge \pi_{bd}(z, \theta)^+$, and thus,

$$U^+_A(z,\theta') = U^+_A(\pi_{bd}(z,\theta'),\theta') \geq U^+_A(\pi_{bd}(z,\theta),\theta').$$

Similarly, if $r^+_{\theta,\theta'}(m)$ is quasi-decreasing in m with respect to \bar{z}^+_0 and convex in m for $m \leq \bar{z}^+_0$,

$$u_A^+(\pi_{bd}(z,\theta'),\theta') \succeq_2 u_A^+(z,\theta') \implies U_A^+(\pi_{bd}(z,\theta'),\theta) \ge U_A^+(z,\theta) = U_A^+(\pi_{bd}(z,\theta),\theta).$$

Hence, $\pi_{bd}(z, \theta')^+ \ge \pi_{bd}(z, \theta)^+$, and thus, again

$$U^+_A(z,\theta') = U^+_A(\pi_{bd}(z,\theta'),\theta') \geq U^+_A(\pi_{bd}(z,\theta),\theta')$$

Similarly, because $U_P^+(z,\theta) = u_A^+(z,\theta) \cdot r_{P,\theta}^+$,

$$u^+_A(\pi_{bd}(z,\theta),\theta) \succeq_1 u^+_A(z,\theta) \implies U^+_P(\pi_{bd}(z,\theta),\theta') \ge U^+_P(z,\theta')$$

if $r_{P,\theta}^+(m)$ is quasi-decreasing in m with respect to \bar{z}_0^+ , and

$$u_A^+(\pi_{bd}(z,\theta),\theta) \succeq_2 u_A^+(z,\theta) \implies U_P^+(\pi_{bd}(z,\theta),\theta') \le U_P^+(z,\theta')$$

if $r^+_{P,\theta}(m)$ is quasi-decreasing in m with respect to \bar{z}^+_0 and convex in m for $m \leq \bar{z}^+_0$

Likewise, the results for the marginal losses can be derived.

F Proofs for the results in Section 6

F.1 Proof of Corollary 1

It suffices to find a sufficient condition for the reducibility of (Θ, Z, U) to single crossing problem under X when Z is defined with the first dominance order. From Theorem 2, the following conditions need to be found. Fix a total order \succeq_{Θ} on Θ . • Increasing MRS

From Proposition 3, this holds if the valuation for the grand bundle $v_{\theta}(b^*)$ is nondecreasing in θ according to the order \succeq_{Θ} : For $\theta' \succeq_{\Theta} \theta$,

$$v_{\theta'}(b^*) \ge v_{\theta}(b^*)$$

• Increasing conflict of interest property

This holds because the principal's preference is type independent.

• Quasi-decreasing $u_A^+(m, \theta)$ and $u_A^-(n, \theta)$ in m and n with respect to \bar{z}_0^+ and \bar{z}_0^- (Property (M_1))

This is equivalent to that $u_A^+(m, \theta)$ is weakly nonincreasing with respect to the cutoff 1, *i.e.*, for every $\theta \in \Theta$ and $b \in B$,

$$v_{\theta}(b^*) \ge v_{\theta}(b).$$

• Quasi-decreasing $r_{\theta,\theta'}^+(m)$, $-r_{\theta,\theta'}^-(n)$, $r_{P,\theta}^+(m)$ and $-r_{P,\theta}^-(n)$ in m and n with respect to \bar{z}_0^+ and \bar{z}_0^- for every $\theta' \succeq_{\Theta} \theta$ (Property (R_1))

This is equivalent to that for every $\theta' \succeq_{\Theta} \theta$, $r_{\theta,\theta'}^+(m)$ is quasi-decreasing in m with respect to the cutoff 1: for every $b \in B$,

$$\frac{v_{\theta}(b^*)}{v_{\theta'}(b^*)} \geq \frac{v_{\theta}(b)}{v_{\theta'}(b)}$$

which is equivalent to $\frac{v_{\theta'}(b^*)}{v_{\theta}(b^*)} \leq \frac{v_{\theta'}(b)}{v_{\theta}(b)}.$

F.2 Proof of Corollary 2

It suffices to show when (Θ, Z, U) is reducible to a single crossing problem under $X := \hat{bd}(Z)$ when Z is defined with the 0th dominance order.. From Theorem 2, fixing a total order \succeq_{Θ} on Θ and the standard order \geq on the set of goods $\{1, \dots, d\}$, the following conditions need to be found:

• Increasing MRS

For Proposition 3, this holds if for every $1 \le m \le d$ and θ, θ' with $\theta' \succeq_{\Theta} \theta$,

$$v_{\theta'}(m) \ge v_{\theta}(m).$$

• Increasing conflict of interest property

This holds because the seller's payoff is type-independent.

• Quasi-decreasing $r^+_{\theta,\theta'}(m)$, $-r^-_{\theta,\theta'}(n)$, $r^+_{P,\theta}(m)$ and $-r^-_{P,\theta}(n)$ in m and n with respect to \bar{z}^+_0 and \bar{z}^-_0 for every $\theta' \succeq_{\Theta} \theta$ (Property (R_1))

This is equivalent to that $r_{\theta,\theta'}^+(m) = \frac{v_{\theta}(m)}{v_{\theta'}(m)}$ is nonincreasing in m: for $m' \ge m$ and $\theta' \succeq_{\Theta} \theta$,

$$\frac{v_{\theta}(m)}{v_{\theta'}(m)} \geq \frac{v_{\theta}(m')}{v_{\theta'}(m')},$$

which is equivalent to $\frac{v_{\theta'}(m')}{v_{\theta'}(m)} \ge \frac{v_{\theta}(m')}{v_{\theta}(m)}$.

It is evident that the same result holds for every permutation on $\{1, \dots, d\}$.

F.3 Proof of Proposition 6

If part is shown in Corollary 2. For only if part, suppose not. That is, MRS between a pair of goods is not monotone according to \succeq_{Θ} . For the ease of exposition, we denote as θ_n the nth lowest type according to the total order \succeq_{Θ} . That is, $\Theta = \{\theta_1, \theta_2, \dots, \theta_N\}$ for some N > 0 and $\theta_N \succeq_{\Theta} \dots \succeq_{\Theta} \theta_2 \succeq_{\Theta} \theta_1$. According to the assumption of strictly increasing valuations, $v_{\theta_n}(k)$ is strictly increasing in n for every $k = 1, \dots, d$. For each good $k = 1, \dots, d$, denote as r_k a vector of the ratios between the valuations of good k of θ_n and θ_{n+1} for every $n = 1, \dots, N-1$:

$$r_k = (\frac{v_{\theta_2}(k)}{v_{\theta_1}(k)}, \frac{v_{\theta_3}(k)}{v_{\theta_2}(k)}, \cdots, \frac{v_{\theta_N}(k)}{v_{\theta_{N-1}}(k)})$$

One can define the lexicographic order on $\{r_1,\cdots,r_d\},$ which can be a partial order.

Also, For $1 \le n \le n' \le N$, denote as $r^{n',n}$ a vector of the ratios between the valuations of good k of $\theta_{n'}$ and θ_n for every $k = 1, \dots, d$.

$$r^{n',n} = (\frac{v_{\theta_{n'}}(1)}{v_{\theta_n}(1)}, \frac{v_{\theta_{n'}}(2)}{v_{\theta_n}(2)}, \cdots, \frac{v_{\theta_{n'}}(d)}{v_{\theta_n}(d)})$$

I define that for $n \leq n' \leq n'' r^{n',n}$ and $r^{n'',n'}$ are well-ordered if for every $1 \leq k \leq k' \leq d$, $(r^{n',n}(k), r^{n'',n'}(k))$ and $(r^{n',n}(k'), r^{n'',n'}(k'))$ are comparable to the lexicographic order \geq_L on \mathbb{R}^2_+ .

Then, the proof proceeds in the following steps.

Step 1 There exist three distinct states θ_n , $\theta_{n'}$ and $\theta_{n''}$ for $n \leq n' \leq n''$ such that $r^{n',n}$ and $r^{n'',n'}$ are not well-ordered.

Proof. First, that MRS between a pair of goods (k_1, k_2) is not monotone according to \succeq_{Θ} means that the lexicographic order over $\{r_1, \cdots, r_d\}$ is not a total order but a partial one, and thus, r_{k_1} and r_{k_2} are not comparable. This implies that there exists $1 < \bar{n} \leq N$ such that for $n < \bar{n}$, $r_{k_1}(n) \geq r_{k_2}(n)$ with a strict inequality for some n and $r_{k_1}(\bar{n}) < r_{k_2}(\bar{n})$. Then, this implies that $r^{\bar{n}-1,1}$ and $r^{\bar{n},\bar{n}-1}$ are not well-ordered because

$$r^{\bar{n},1}(k_1) > r^{\bar{n},1}(k_2), \ r^{\bar{n}+1,\bar{n}}(k_1) < r^{\bar{n}+1,\bar{n}}(k_2).$$

For the ease of exposition, I call the three states in Step 1, $\theta_3 \succeq_{\Theta} \theta_2 \succeq_{\Theta} \theta_1$. Then, I construct a three point distribution μ over the three states under which every optimal under which IC mechanism is not an upgrade pricing.

First, I relabel the indexes of good so that

1. $r^{3,2}(1) \le r^{3,2}(2) \le \dots \le r^{3,2}(d)$, and 2. if $r^{3,2}(k) = r^{3,2}(k+1)$ for some $k, r^{2,1}(k) \le r^{2,1}(k+1)$.

Then, I define the following set D of the indexes:

$$D := \{k | \{k' | r^{2,1}(k') \le r^{2,1}(k)\} \subsetneq \{k' | r^{3,2}(k') \le r^{3,2}(k)\}\}.$$

Step 2 $D \neq \emptyset$

Proof. First, consider the largest index k satisfying that there exists k' s.t.

$$r^{2,1}(k) < r^{2,1}(k') \text{ and } r^{3,2}(k) > r^{3,2}(k').$$
 (35)

Due to Step 1, there exists an index k that satisfies (35). Also, for the largest k,

$$\{k'|r^{2,1}(k') \leq r^{2,1}(k)\} \neq \{k'|r^{3,2}(k') \leq r^{3,2}(k)\}$$

Hence, if the largest k is not an element of D, this means that there exists an k'' s.t.

$$r^{2,1}(k'') \leq r^{2,1}(k) \text{ but } r^{3,2}(k'') > r^{3,2}(k).$$

Then, combined with (35), it implies that

$$r^{2,1}(k'') < r^{2,1}(k') \mbox{ and } r^{3,2}(k'') > r^{3,2}(k')$$

which contradicts with the assumption that k is the largest index satisfying (35) for some k'. That is, such k is an element of D. \Box

Letting $\hat{r}^{3,2} := \max_{k \in D} r^{3,2}(k)$, the set of the indexes that has the value of $r^{3,2}$ equal to $\hat{r}^{3,2}$ can be written as

$$\{k^*,k^*+1,\cdots,k^*+l^{3,2}\}=\{k|r^{3,2}(k)=\hat{r}^{3,2}(k)\}$$

for some $k^* = \{1, \dots, d\}$ and $l^{3,2} \ge 0$. According to the definition of D, this also implies that for every $k \le k^* + l^{3,2}$,

$$r^{2,1}(k) \le r^{2,1}(k^* + l^{3,2} + 1) \le \dots \le r^{2,1}(n)$$
(36)

and there exist $l^{2,1} \leq l^{3,2}$ and a permutation σ on $\{1, \dots, k^* - 1\}$ with some $k_1^* \leq k^* - 1$ s.t. for every $k \leq k_1^* < k'$,

$$r^{2,1}(k^*) = \dots = r^{2,1}(k^* + l^{2,1}) < r^{2,1}(k^* + l^{2,1} + 1) \dots \le r^{2,1}(k^* + l^{3,2})$$
(37)

$$r^{2,1}(\sigma(1)) \le \dots \le r^{2,1}(\sigma(k_1^*)) \le r^{2,1}(k^*) < r^{2,1}(\sigma(k_1^*+1)) \le \dots \le r^{2,1}(\sigma(k^*-1))$$
(38)

Now, I consider the following distribution $\mu = (\mu_1, \mu_2, \mu_3)$ over $\{\theta_1, \theta_2, \theta_3\}$ to show that upgrade pricing is not optimal for μ :

$$\mu_1 v_1(k^*) - \mu_2(v_2(k^*) - v_1(k^*)) > 0 > \mu_1 v_1(\sigma(k_1^* + 1)) - \mu_2(v_2(\sigma(k_1^* + 1)) - v_1(\sigma(k_1^* + 1)))$$

$$(39)$$

$$\mu_2 v_2(k^* - 1) - \mu_3(v_3(k^* - 1) - v_2(k^* - 1)) > 0 > \mu_2 v_2(k^*) - \mu_3(v_3(k^*) - v_2(k^*)).$$
(40)

Hence, if only the IR constraints and the local downward IC constraints are required, the optimal mechanism \tilde{h}^* will assign type θ_1 a bundle $B_1 = \{\sigma(1), \dots, \sigma(k_1^*), k^*, \dots, k^* + l^{2,1}\}$, type θ_2 a bundle $B_2 = \{1, \dots, k^* - 1\}$ and type θ_3 the grand bundle $B_3 = b^*$ with the prices that make all the local downward ICs and the IR for type θ_1 binding.

To show that upgrade pricing is not optimal for the type distribution μ , I consider the following two cases:

(*Case 1*) The valuation for each good is sufficiently convex in type:

$$\sum_{k \in B_2} v_3(k) - \sum_{k \in B_1} v_3(k) \ge \sum_{k \in B_2} v_2(k) - \sum_{k \in B_1} v_2(k) \tag{41}$$

(Case 2) The valuation for each good is not sufficiently convex in type:

$$\sum_{k \in B_2} v_3(k) - \sum_{k \in B_1} v_3(k) < \sum_{k \in B_2} v_2(k) - \sum_{k \in B_1} v_2(k)$$
(42)

In the following steps, I show that optimal LIC and IC mechanisms are equivalent in each of the above two cases. Moreover, in the first case, the optimal IC mechanism has only the local downward IC constraints binding but it is not an upgrade pricing. In the second case, it has all the downward IC constraints binding, which also means that upgrade pricing is not optimal due to the strictly monotone valuation or the resulting strict single crossing property.

(Step 3) In Case 1, \tilde{h}^* satisfies all the constraints. That is, it is the optimal IC mechanism, which means that upgrade pricing is not optimal.

Proof. First, note that because of the strictly type-increasing valuations, the IR constraint for θ_1 and the IC constraints imply the IR constraints for the other types. Hence, it suffices to show that all the IC constraints are satisfied under \tilde{h}^* . let's first see the local upward IC constraints. The upward IC between θ_2 and θ_3 holds because the downward IC between them is binding and the strictly monotone valuation implies that

$$\sum_{k \in b^*} v_3(k) - \sum_{k \in B_2} v_3(k) \geq \sum_{k \in b^*} v_2(k) - \sum_{k \in B_2} v_2(k).$$

The upward IC between θ_1 and θ_2 holds because the downward IC between them is binding and θ_2 has a higher relative valuation for goods in B_2/B_1 compared to goods in B_1/B_2 than θ_1 .

Then, the remaining IC constraints are those between θ_1 and θ_3 . First, because of (41), which implies that θ_3 has a stronger preference for B_2 over B_1 then θ_2 . Also, because θ_2 has a stronger preference for $B_3 = b^*$ over B_2 than θ_1 due to the type-monotone valuations. This implies that the agent's preference over the three bundles, b^*, B_1 , and B_2 , has the single crossing property, and thus, the local IC constraints imply the non-local IC constraints.

Hence, all the IC constraints are satisfied under \tilde{h}^* , which means that \tilde{h}^* is the optimal IC mechanism. Moreover, due to the definition of k^* , $B_1 \not\subseteq B_2$, and thus, upgrade pricing is not optimal. \Box

(Step 4) In *Case 2*, the optimal LIC mechanisms are the optimal IC mechanisms, and they have all the downward IC constraints binding. Hence, because of the strictly type-monotone valuations, the optimal IC mechanisms are not upgrade pricing.

Proof. Consider an optimal LIC mechanism h^* . It is evident that (1) h^* assigns θ_3 the grand bundle, and (2) the IR constraint for θ_1 and the downward IC constraint between θ_2 and θ_1 are binding. Moreover, at least one of the other two downward IC constraints, the constraint for θ_1 and θ_3 and the constraint for θ_2 and θ_3 , is binding. In the former case, this contradicts with the optimality of h^* because the seller can extract more rent by selling more goods to θ_2 . In the latter case, suppose that the downward IC between θ_3 and θ_1 is not binding. Then, h^* has to assign θ_2 the bundle B_2 and θ_1 a bundle larger than B_1 because if not, the seller can extract more rents by changing bundles assigned to θ_2 and θ_1 without violating any downward IC constraints. Then, (42) implies that downward IC between θ_3 and θ_1 .

It remains to show that h^* satisfies all the constraints. Again, it suffices to show that the upward IC constraint between θ_2 and θ_1 holds. Again, h^* has to assign θ_2 a bundle larger than B_2 and θ_1 a bundle smaller than B_1 because if not, the seller can extract more rents by changing bundles assigned to θ_2 and θ_1 without violating any downward IC constraints. The latter also implies that h^* has to assign θ_2 a bundle smaller than $\{1, \dots, k^* + l^{2,1}\} (\supseteq B_1)$, because otherwise the strictly type monotone valuation implies that the downward IC between θ_3 and θ_1 is not binding. Due to the same logic, h^* has to assign θ_1 a bundle larger than $\{\sigma(1), \dots, \sigma(k_1^*)\} (\subseteq B_2)$. That is, denoting as B_1^* and B_2^* the bundles assigned to θ_1 and θ_2 by h^* ,

$$\begin{aligned} \{\sigma(1), \cdots, \sigma(k_1^*)\} \subseteq B_1^* \subseteq B_1 &= \{\sigma(1), \cdots, \sigma(k_1^*), k^*, \cdots, k^* + l^{2,1} \} \\ \{1, \cdots, k^* - 1\} &= B_2 \subseteq B_2^* \subseteq \{1, \cdots, k^* + l^{2,1} \} \end{aligned}$$

Hence, the upward IC between θ_1 and θ_2 holds because the downward IC between them is binding and θ_2 has a higher relative valuation for goods in B_2^*/B_1^* compared to goods in B_1^*/B_2^* than θ_1 .

F.4 Proof of Corollary 4

First, I show when (Θ, Z, U) is reducible to a single crossing problem under $X = \hat{bd}(Z)$. According to Theorem 2, the lower contour set Z with the second order dominance order, the following conditions need to be obtained given a total order \succeq_{Θ} on Θ . According to Proposition 10, (\hat{S}_2) instead of the increasing conflict of interest will be shown for the optimality of no screening. Here, I use the standard order \geq as the total order.

• Increasing MRS

This holds if $u_A(c,\theta) = (1 - G(c - \theta))(\mathbb{E}[\tilde{c}|\tilde{c} \ge c - \theta] - (c - \theta))$ is nondecreasing in θ , which is true because its first order derivative $\frac{\partial}{\partial \theta}u_A(c,\theta)$ is equal to $1 - G(c - \theta) \ge 0$.

• Quasi-decreasing and convex $u_A^+(m,\theta)$ and $u_A^-(n,\theta)$ in m and n with respect to \overline{z}_0^+ and \overline{z}_0^- . (Property (M_2))

This again holds because $\frac{\partial}{\partial c}u_A(c,\theta)=-(1-G(c-\theta))\leq 0\leq g(c-\theta)=\frac{\partial^2}{\partial c^2}u_A(c,\theta).$

- Quasi-decreasing and convex $r^+_{\theta,\theta'}(c^+)$, $-r^-_{\theta,\theta'}(c^-)$, $r^+_{P,\theta}(c^+)$ and $-r^-_{P,\theta}(c^-)$ in c^+ and c^- with respect to \bar{z}^+_0 and \bar{z}^-_0 for every $\theta' \succeq_{\Theta} \theta$ (Property (R_2))
- Quasi-decreasing and convex $r^+_{\theta,\theta'}(c^+)$, $-r^-_{\theta,\theta'}(c^-)$, $r^+_{P,\theta}(c^+)$ and $-r^-_{P,\theta}(c^-)$ in c^+ and c^- with respect to \bar{z}^+_0 and \bar{z}^-_0 for every $\theta' \succeq_{\Theta} \theta$ (Property (R_2))

As in footnote 17, I use a stronger version of Property (R_2) when \bar{z} has the full support and $C^- = \emptyset$:

 $(\hat{R_2}) \ \frac{u_P^+(c,\theta)}{u_A^+(c,\theta)}, \ \frac{\frac{\partial}{\partial c}u_P^+(c,\theta)}{\frac{\partial}{\partial c}u_A^+(c,\theta)}, \ , -\frac{\frac{\partial}{\partial c}u_A^+(c,\theta)}{u_A^+(c,\theta)}, \ \text{and} - \frac{\frac{\partial^2}{\partial c\partial \theta}u_A^+(c,\theta)}{\frac{\partial}{\partial c}u_A^+(c,\theta)} \ \text{are nonincreasing in } c.$

Because

$$\begin{split} & \frac{u_P^+(c,\theta)}{u_A^+(c,\theta)} = -\frac{\theta}{\mathbb{E}[\tilde{s}|\tilde{s} \ge c-\theta] - (c-\theta)} \\ & \frac{\frac{\partial}{\partial c} u_P^+(c,\theta)}{\frac{\partial}{\partial c} u_A^+(c,\theta)} = -\theta \frac{g}{1-G}(c-\theta) \\ & -\frac{\frac{\partial}{\partial c} u_A^+(c,\theta)}{u_A^+(c,\theta)} = -\frac{1}{\mathbb{E}[\tilde{s}|\tilde{s} \ge c-\theta] - (c-\theta)} \\ & -\frac{\frac{\partial^2}{\partial c\partial \theta} u_A^+(c,\theta)}{\frac{\partial}{\partial c} u_A^+(c,\theta)} = -\frac{g}{1-G}(c-\theta), \end{split}$$

all of these nonincreasingness hold if the hazard ratio $\frac{g}{1-G}(s)$ is nondecreasing in s. Especially, $\mathbb{E}[\tilde{s}|\tilde{s} \ge c - \theta] - (c - \theta)$ are also nonincreasing in c

• (\hat{S}_2)

It suffices to find when the single crossing property of $U|_{X\times\Theta}$ with $(\geq,\succeq_{\Theta}^{D})$ holds. This holds because $u_A(c,0) = u_P(c,0)$ for every c and

$$\frac{\partial}{\partial \theta} u_A(c,\theta) = 1 - G(c-\theta) \geq 0 \geq -\theta g(c-\theta) = \frac{\partial}{\partial \theta} u_P(c,\theta).$$

Proof of Corollary 3 **F.5**

I first show when (Θ, Z, U) is reducible to a single crossing problem under $X = \hat{bd}(Z)$ by using a stronger version of Theorem 2. Proposition 7 shows that increasing conflict of interest can be replaced with (10). Then, because Z is the lower contour set Z of 1 with respect to the 0th dominance order, only the following conditions need to be found for a given total order \succeq_{Θ} on Θ :

• Increasing MRS

This holds if for every s' > 0 > s, $\frac{u_A^+(s',\theta)}{u_A^-(-s,\theta)} = -\frac{f(s',\theta)u(s')}{f(s,\theta)u(s)}$ is nondecreasing in θ according to the order \succeq_{Θ} . Because u(s') > 0 > u(s), it holds if $\frac{f(s',\theta)}{f(s,\theta)}$ is nondecreasing in θ , that is, $\frac{f(s,\theta')}{f(s,\theta)} \leq \frac{f(s',\theta')}{f(s',\theta)}$ for every $\theta' \succeq_{\Theta} \theta$.

• Quasi-decreasing $r^+_{\theta,\theta'}(c^+), -r^-_{\theta,\theta'}(c^-), r^+_{P,\theta}(c^+)$ and $-r^-_{P,\theta}(c^-)$ in c^+ and c^- with respect to \bar{z}_0^+ and \bar{z}_0^- for every $\theta' \succeq_{\Theta} \theta$ (Property (R_1))

For every $\theta' \succeq_{\Theta} \theta$,

$$r_{\theta,\theta'}^+(c^+) := \frac{u_A^+(c^+,\theta)}{u_A^+(c^+,\theta')} = \frac{f(c^+,\theta)}{f(c^+,\theta')}: \text{ nonincreasing in } c^+$$

$$r^-_{\theta,\theta'}(c^-) := \frac{u^-_A(c^-,\theta)}{u^-_A(c^-,\theta')} = \frac{f(-c^-,\theta)}{f(-c^-,\theta')}: \text{ nondecreasing in } c^-$$

$$r_{P,\theta}^+(c^+) := \frac{u_P^+(c^+,\theta')}{u_A^+(c^+,\theta)} = \frac{v(c^+)}{u(c^+)}: \text{ nonincreasing in } c^+$$

$$r^-_{P,\theta}(c^-):=\frac{u^-_P(c^-,\theta')}{u^-_A(c^-,\theta)}=\frac{v(-c^-)}{u(-c^-)}$$
: nondecreasing in c^-

The first two conditions with the above condition for increasing MRS are equivalent to (8) while the last two are equivalent to (9).

Moreover, due to the single crossing condition of (Θ, X, U) , every IC mechanism h under X has to satisfy that for $\theta' \succeq_{\Theta} \theta$,

$$h(\theta) \subseteq h(\theta'),\tag{43}$$

which means that h takes the form of the nested acceptance interval structure. Therefore, the optimal IC mechanism under X, which is also an optimal IC mechanism under Z, also satisfies the nested acceptance interval structure.

Lastly, I need to show the implementability of such IC mechanism h through an information structure. Because $U_A(h(\theta'), \theta') \ge U_A(h(\theta), \theta')$ and $U_A(h(\theta), \theta) \ge U_A(h(\theta'), \theta)$ for $\theta' \succeq_{\Theta} \theta$, the single crossing property implies with (43) that for every $\theta'' \succeq_{\Theta} \theta'$,

$$U_A(h(\theta'),\theta'') \geq U_A(h(\theta),\theta'')$$

and for every $\theta \succeq_{\Theta} \theta''$,

$$U_A(h(\theta),\theta'') \geq U_A(h(\theta'),\theta'')$$

Hence, consider an information structure under which S is partitioned into

$$\{h(\theta_n)\backslash h(\theta_{n-1})|n=1,\cdots,N+1\}$$

where $\Theta = \{\theta_N, \cdots, \theta_1\}$ with $\theta_N \succeq_{\Theta} \cdots, \succeq_{\Theta} \theta_1$, $h(\theta_0) \equiv \emptyset$, and $h(\theta_{N+1}) \equiv S$. Then, for every $n' \leq n \leq n''$

$$U_A(h(\theta_{n'}) \backslash h(\theta_{n'-1}), \theta_n) \geq U_A(\emptyset, \theta_n) \geq U_A(h(\theta_{n''}) \backslash h(\theta_{n''-1}), \theta_n)$$

Therefore, each agent with type θ_n will choose the acceptance set $h(\theta_n)$, which implies that every nested interval structure of acceptance sets from an IC mechanism h can be implemented by the information structure.

G Weakly increasing conflict of interest property

G.1 General result

In Section 4.4, I provided a sufficient condition for the equivalence between optimal LIC and IC mechanisms in single crossing screening problem. Basically, the condition says that for a set of pairs of allocations that can be used by an optimal LIC mechanism with violating an upward IC constraint, the principal is less likely to have the same preference with the agent as type goes up. In fact, this condition can be weakened for some of the pairs of allocations. First, I define the following allocation subset X_s .

Definition 12 In a single crossing screening problem (Θ, X, U) , X_s is the set of allocations that satisfies the following:

For every $x \in X_s$ and every $\epsilon > 0$, there exists an $x'' \in X$ satisfying that $|x - x''| < \epsilon$ and for every $\theta \in \Theta$

$$U_P(x'',\theta) > U_P(x,\theta) \text{ and } U_A(x,\theta) \ge U_A(x'',\theta).$$

In other words, X_s is the set of allocation x such that there exists a direction v with which moving x decreases the agent's payoff but increases the principal's payoff regardless of the agent's type. For example, in single good monopoly problem where the seller maximizes the expected revenue, an increase in price is the direction of the allocation that leads to the decrease in the buyer's payoff and the increase in the seller's payoff. Hence, $X_s = X$.

Such allocation can be assigned to a type of agent by an optimal LIC mechanism only when a downward IC constraint of the type of the agent is binding. If not, the principal's payoff can be strictly increased without violating any downward IC constraint. Hence, the property of increasing conflict of interest can be weakened for $x \in X_s$.

Definition 13 A screening problem (Θ, X, U) has weakly increasing conflict of interest with a total order \succeq_{Θ} on Θ if the followings hold together:

• When $x \in X_s$, for every $x' \in X$ and $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$,

$$\begin{cases} U_A(x,\theta) = U_A(x',\theta) \\ U_A(x,\theta') \ge U_A(x',\theta') \implies U_P(x,\theta') \ge U_P(x',\theta'). \\ U_P(x,\theta) \ge U_P(x',\theta) \end{cases}$$

• When $x \notin X_s$, for every $x' \in X$ and $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$,

$$\begin{cases} U_A(x,\theta) \ge U_A(x',\theta) \\ U_A(x,\theta') \ge U_A(x',\theta') \\ U_P(x,\theta) \ge U_P(x',\theta) \end{cases} \implies U_P(x,\theta') \ge U_P(x',\theta').$$

Proposition 11 Suppose that (Θ, X, U) is a single crossing problem and has weakly increasing conflict of interest with $(\succeq_X, \succeq_{\Theta})$. Then, for all type distributions $\mu \in \Delta\Theta$, every optimal IC mechanism under X is an optimal LIC mechanism under X with \succeq_{Θ} .

Proof. The proof is same as the proof of Proposition 2 up to Step 3. Again, given an agentoptimal optimal LIC mechanism h under X, let n^* be the smallest n that does not satisfy the local upward IC constraint between θ_n and θ_{n+1} . The only difference is that the local downward IC constraint between θ_{n^*} and θ_{n^*+1} has to be binding if $h(\theta_{n^*+1}) \in X_s$. If not, the principal's payoff is strictly increased by replacing $h(\theta_{n+1})$ with another allocation $x'' \in X$ s.t.

- x'' is more preferred to $h(\theta_{n+1})$ by the principal and less preferred by the agent regardless of the agent's type, and
- $|x'' h(\theta_{n+1})|$ is sufficiently small so that the agent with type θ_{n+1} still prefers x'' to $h(\theta_n)$.

First, the existence of such x'' is guaranteed by $h(\theta_{n+1}) \in X_s$. Also, due to (21) in the proof of Proposition 2, the second condition also means that he prefers x'' to $h(\theta_m)$ for every $m \leq n$. Hence, the change in allocation increases the principal's payoff without violating any downward IC constraint, which contradicts with the optimality of h. Therefore, in this case when $h(\theta_{n+1}) \in X_s$, the local downward IC constraint is binding, and thus, (23) is replaced with the binding constraint:

$$U_A(h(\theta_{n^*+1}),\theta_{n^*+1}) = U_A(h(\theta_{n^*}),\theta_{n^*+1}).$$

Hence, the first condition in the weakly increasing conflict of interest property is enough to show a contradiction. \blacksquare

The weakly increasing conflict of interest property is equivalent to the increasing conflict of interest property if $X_s = \emptyset$. Moreover, the former is a weaker condition than the latter because the condition for $x \in X_s$ is weaker than the condition for $x \notin X_s$ in that one only needs to consider x' that gives the same payoff to type θ as x does. Especially, in a strict single crossing problem (Θ, X, U) with $(\succeq_{\Theta}, \succeq_X)$, this also implies $x' \succeq_X x^{.37}$

Remark 4 When money transfers are allowed, $X_s = X$. Moreover, the first condition of weakly increasing conflict of interest property can be even further weakened by replacing the principal's payoff with the social welfare in its last inequality because the principal can allocate

 $^{^{37}}$ A screening problem is a strict single crossing problem if it is a single crossing problem and all the conditions in Definition 1 also hold for strict inequalities

the welfare to the players however she wants: For every $x, x' \in X$ and $\theta, \theta' \in \Theta$ with $\theta \succeq_{\Theta} \theta'$

$$\begin{cases} U_A(x,\theta) = U_A(x',\theta) \\ U_A(x,\theta') \ge U_A(x',\theta') \\ U_P(x,\theta) \ge U_P(x',\theta) \end{cases} \implies U_P(x,\theta') + U_A(x,\theta') \ge U_P(x',\theta') + U_A(x',\theta').^{38}$$

Proof. As in Proposition 11, the proof is same up to Step 3 of Proposition 2, and hence, it suffices to check that for an agent-optimal optimal LIC mechanism h^* under X this property has to hold when $\theta = \theta_{n+1}$, $\theta' = \theta_n$, $x = h(\theta_{n+1})$ and $x' = h(\theta_n)$. If this does not hold, h^* is not agent-optimal: By replacing $h(\theta_n)$ with $h(\theta_{n+1}) - t$ for a transfer t > 0 with $U_A(h(\theta_n), \theta_n) = U_A(h(\theta_{n+1}) - t, \theta_n)$, without violating any downward IC constraint either the principal's payoff can be strictly increased or the variance among the allocations to the highest types can get smaller.

Therefore, the main theorem still holds even if increasing conflict of interest property is replaced with its weak version.

Theorem 4 Suppose that there exist a total order \succeq_{Θ} on Θ and an allocation subset $X \subseteq Z$ with a total order \succeq_X s.t. with $(\succeq_X, \succeq_{\Theta})$

- X is an incentive contract curve, and
- (Θ, X, U) is a single crossing one and has weakly increasing conflict of interest.

Then, (Θ, Z, U) is reducible to a single crossing problem under X.

G.2 Weakly increasing conflict of interest in the setting of Section 5

Assuming the additional structure as in Section 5, one can obtain the following sufficient conditions for the weakly increasing conflict of interest property.

Proposition 12 Suppose that type set Θ has increasing MRS between positive and negative components with \succeq_{Θ} . Also, for some $0 \leq \bar{n} \leq N$, the principal's marginal benefits from all positive components are positive while her marginal loss from negative component -n is negative for $n \leq \bar{n}$ and otherwise positive: For every θ , m and $n \leq \bar{n} < n'$,

$$u_P^+(m,\theta)>0,\;u_P^-(n,\theta)<0\leq u_P^-(n',\theta).$$

 $^{^{38}}$ This is equivalent to the single crossing property of the social welfare assumed in Yang (2021).

Then, for $X = \hat{bd}(Z)$, the weakly increasing conflict of interest property holds if for all m, $n > \bar{n}$ and $\theta \succeq_{\Theta} \theta'$,

$$\frac{u_A^+(m,\theta')}{u_A^-(n,\theta')} \leq \frac{u_P^+(m,\theta')}{u_P^-(n,\theta')} \leq \frac{u_P^+(m,\theta)}{u_P^-(n,\theta)}$$

Proof. Let $C_P^+ := \{1, \dots, \bar{n}\} \subseteq C^-$. The principal has positive marginal payoff from every negative component in C_P^+ . Then, it is straightforward to show that

$$X_s \supseteq \tilde{X_s} := \{ x \in \hat{bd}(Z) | (\bar{z}^+, \bar{z}^-|_{C_P^+}) \succeq_{bd} x \text{ and } \bar{z}^-(n) > x^-(n) \text{ for some } 1 \le n \le \bar{n} \}$$

because for the allocation $x \in \tilde{X}_s$, the principal's payoff can be strictly increased with decreasing all types of the agent's payoffs by raising the value of $x^-(n)$. Now, consider the following two cases and show that the weakly increasing conflict of interest holds.

• $x \in \tilde{X_s}$

First, $U_A(x,\theta) = U_A(x',\theta)$ implies that either $x \ge x'$ or $x' \ge x$ because otherwise the equality cannot hold. Also, if $x \ge x'$, then $U_P(x,\theta') \ge U_P(x',\theta')$ because $x^-(n) \ge 0$ only for $n \in C_P^+$ and for every positive component and every negative component in C_P^+ the principal have nonnegative marginal payoffs.

It only remains to show that the first part of (\hat{S}) holds when $x' \ge x$. Because

$$\frac{u_A^+(m,\theta)}{u_A^-(n,\theta)} \leq \frac{u_P^+(m,\theta)}{u_P^-(n,\theta)},$$

 $U_A(x,\theta) = U_A(x',\theta)$ implies $U_P(x',\theta) \ge U_P(x,\theta)$ due to the same logic of single crossing property. Moreover, because $\bar{z}^-(n) > x^-(n)$ for some $1 \le n \le \bar{n}$, this inequality has to strictly hold, which does not satisfy the assumption of the first part of the weakly increasing conflict of interest property.

• $x \notin \tilde{X_s}$

First, because $U_A(x,\theta) \ge U_A(x',\theta)$ in the second part of the weakly increasing conflict of interest property (that is, the increasing conflict of interest), it cannot be the case that $x^+ \le x'^+$ and $x^- \ge x'^-$. Also, in the opposite case where $x^+ \ge x'^+$ and $x^- \le x'^-$, $U_P(x,\theta') \ge U_P(x',\theta')$ always holds, which means that the increasing conflict of interest always holds. Hence, the remaining two cases are $(x^+, x^-) \ge (x'^+, x'^-)$ and $(x^+, x^-) \le (x'^+, x'^-)$. $\begin{array}{l} - \ (x^+,x^-) \geq (x'^+,x'^-) \\ \text{Because } \ \frac{u_A^+(m,\theta')}{u_A^-(n,\theta')} \leq \frac{u_P^+(m,\theta')}{u_P^-(n,\theta')}, \ \text{the single crossing property implies the following:} \end{array}$

$$U_A(x,\theta') \geq U_A(x',\theta') \implies U_P(x,\theta') \geq U_P(x',\theta').$$

Thus, the increasing conflict of interest holds.

-
$$(x^+, x^-) \leq (x'^+, x'^-)$$

Because $\frac{u_P^+(m, \theta')}{u_P^-(n, \theta')} \leq \frac{u_P^+(m, \theta)}{u_P^-(n, \theta)}$, again the single crossing property implies the following:

$$U_P(x,\theta) \ge U_P(x',\theta) \implies U_P(x,\theta') \ge U_P(x',\theta').$$

Thus, the increasing conflict of interest holds.

G.3 Proof of Proposition 7

 $\begin{array}{l} \text{Proposition 7 is a corollary of the continuous version of Proposition 12. Because for $s \leq s_0 \leq 0 \leq s'$ $\frac{u_A^+(s',\theta')}{u_A^-(-s,\theta')} = -\frac{f(s',\theta')}{f(s,\theta')} \frac{u(s')}{u(s)}, $\frac{u_P^+(s',\theta')}{u_P^-(-s,\theta')} = -\frac{f(s',\theta')}{f(s,\theta')} \frac{v(s')}{v(s)}$, $ $\frac{v(s')}{v(s)}$, $\frac{v(s')}{v(s)}$, $\frac{v(s')}{v(s)} = -\frac{f(s',\theta')}{f(s,\theta')} \frac{v(s')}{v(s)}$, $\frac{v(s')}{v(s)} = -\frac{f(s',\theta')}{v(s)} \frac{v(s')}{v(s)}$, $\frac{v(s')}{v(s)} = -\frac{f(s',\theta')}{v(s)}$, $\frac{v(s')}{v(s)} = -\frac{f(s',\theta')}{v$

$$\frac{u_A^+(s',\theta')}{u_A^-(-s,\theta')} \leq \frac{u_P^+(s',\theta')}{u_P^-(-s,\theta')} \leq \frac{u_P^+(s',\theta)}{u_P^-(-s,\theta)}$$

implies that $\frac{u(s')}{u(s)} \geq \frac{v(s')}{v(s)}$ and $\frac{f(s',\theta')}{f(s,\theta')} \geq \frac{f(s',\theta)}{f(s,\theta)}$. The latter comes from the single crossing condition of (Θ, X, U) .

References

- Alonso, R. and N. Matouschek (2008). Optimal delegation. The Review of Economic Studies 75(1), 259–293.
- Araujo, A. and H. Moreira (2010). Adverse selection problems without the spence–mirrlees condition. *Journal of Economic Theory* 145(3), 1113–1141.
- Armstrong, M. (1996). Multiproduct nonlinear pricing. Econometrica 64(1), 51–75.
- Bergemann, D., A. Bonatti, A. Haupt, and A. Smolin (2021). The optimality of upgrade pricing.In International Conference on Web and Internet Economics, pp. 41–58. Springer.

- Cai, Y., N. R. Devanur, and S. M. Weinberg (2016). A duality based unified approach to bayesian mechanism design. In *Proceedings of the forty-eighth annual ACM symposium on Theory of Computing*, pp. 926–939.
- Carroll, G. (2012). When are local incentive constraints sufficient? *Econometrica* 80(2), 661–686.
- Carroll, G. (2017). Robustness and separation in multidimensional screening. *Economet*rica 85(2), 453–488.
- Chen, C.-H., J. Ishida, and W. Suen (2022). Signaling under double-crossing preferences. *Econometrica* 90(3), 1225–1260.
- Daskalakis, C., A. Deckelbaum, and C. Tzamos (2017). Strong duality for a multiple-good monopolist. *Econometrica* 85, 735–767.
- Eső, P. and B. Szentes (2007). Optimal information disclosure in auctions and the handicap auction. *The Review of Economic Studies* 74(3), 705–731.
- Guo, Y. and E. Shmaya (2019). The interval structure of optimal disclosure. *Econometrica* 87(2), 653–675.
- Haghpanah, N. and J. Hartline (2021). When is pure bundling optimal? The Review of Economic Studies 88(3), 1127–1156.
- Hart, S. and N. Nisan (2019). Selling multiple correlated goods: Revenue maximization and menu-size complexity. *Journal of Economic Theory* 183, 991–1029.
- Hörner, J. (2008). Signalling and screening. The New Palgrave Dictionary of Economics, 15.
- Manelli, A. M. and D. R. Vincent (2006). Bundling as an optimal selling mechanism for a multiple-good monopolist. *Journal of Economic Theory* 127(1), 1–35.
- Matthews, S. and J. Moore (1987). Monopoly provision of quality and warranties: An exploration in the theory of multidimensional screening. *Econometrica* 55(2), 441–467.
- Melumad, N. D. and T. Shibano (1991). Communication in settings with no transfers. *The RAND Journal of Economics* 22(2), 173–198.
- Milgrom, P. and C. Shannon (1994). Monotone comparative statics. *Econometrica* 62(1), 157–180.

- Moore, J. (1984). Global incentive constraints in auction design. *Econometrica* 52(6), 1523–1535.
- Mussa, M. and S. Rosen (1978). Monopoly and product quality. *Journal of Economic the*ory 18(2), 301–317.
- Pavlov, G. (2011). Optimal mechanism for selling two goods. The BE Journal of Theoretical Economics 11(1).
- Riley, J. and R. Zeckhauser (1983). Optimal selling strategies: When to haggle, when to hold firm. *The Quarterly Journal of Economics* 98(2), 267–289.
- Rochet, J.-C. and P. Choné (1998). Ironing, sweeping, and multidimensional screening. *Econo*metrica 66(4), 783–826.
- Rochet, J.-C. and L. A. Stole (2003). The economics of multidimensional screening. *Econometric Society Monographs* 35, 150–197.
- Schottmüller, C. (2015). Adverse selection without single crossing: Monotone solutions. Journal of Economic Theory 158, 127–164.
- Strassen, V. (1965). The existence of probability measures with given marginals. The Annals of Mathematical Statistics 36(2), 423–439.
- Tanner, N. (2018). Optimal delegation under unknown bias: The role of concavity. FRB Boston Risk and Policy Analysis Unit Paper No. RPA, 18–1.
- Thanassoulis, J. (2004). Haggling over substitutes. *Journal of Economic theory* 117(2), 217–245.
- Yang, F. (2021). Costly multidimensional screening. arXiv preprint arXiv:2109.00487.